

Information Management School



Mestrado em Gestão de Informação

Master Program in Information Management

THE FACTORS INFLUENCING THE ADOPTION OF MENTAL HEALTH APPLICATIONS

Alicia Schmidt

Dissertation presented as partial requirement for obtaining the Master's degree in Information Management with a specialization in Marketing Intelligence

NOVA Information Management School Instituto Superior de Estatística e Gestão de Informação

Universidade Nova de Lisboa

ABSTRACT

Mental illnesses are a growing issue nowadays. According to the World Health Organization, the gap between the need for treatment for mental disorders and the accessibility of treatment is widening. One potential solution to making treatment more accessible is mental health apps. In recent years, mental health apps have been essential tools for providing healthcare services at an affordable cost. Despite the effectiveness and benefits of new technologies, the uptake of these apps remains a challenge, especially in Europe compared to the US. This study aims to investigate the factors that influence the adoption of mental health apps. To this end, a conceptual framework based on the UTAUT2 and the HBM was developed, which was tested using a quantitative study. For operationalization, 309 participants aged between 18 and 70 years old were collected through an online questionnaire. The results show that all of the tested factors impact behavioral intention. The determinants that stand out are performance expectancy, social influence, hedonic motivation, and cues to action, which significantly affect the behavioral choice to use a mental health application. The findings lead to implications for promoting and developing greater adoption of mental health apps.

KEYWORDS

Mental health applications; Mental health apps; Technology adoption; UTAUT2; HBM

INDEX

1.	Introduction	7
	1.1. Research Problem	7
	1.2. Context of the study	8
	1.3. Research purpose and objectives	9
	1.4. Research significance	9
	1.5. Research question	.10
	1.6. Outline of the report	.10
2.	Literature review	.13
	2.1. Overview	.13
	2.2. Mental health disorders	.13
	2.2.1. Definition	.13
	2.2.2.Evolution	.14
	2.2.3. Influencing Factors	.15
	2.3. Mental health applications	.22
	2.3.1. Definition	.22
	2.3.2. History	.23
	2.3.3. Market outlook	.25
	2.3.4. Benefits	.25
	2.3.5. Drwabacks	.26
	2.4. Theoretical frameworks	.28
	2.4.1.Technology acceptance model	.28
	2.4.2. Extended technology acceptance model	.30
	2.4.3. Unified theory of acceptance and use of technology	.31
	2.4.4. Unified theory of acceptance and use of technology 2	.33
	2.4.5.Health belief model	.34
3.	Methodology	.38
	3.1. Overview	.38
	3.2. Research objectives	.38
	3.3. Conceptual framework and research hypothesis	.38
	3.4. Participants	.45
	3.5. Procedure	.46
	3.6. Measures	.47
	3.6.1.Questionnaire design	.47

	3.6.2. Measurement scales	.48
	3.7. Data collection	.52
4.	Results	.54
	4.1. Overview	.54
	4.2. Psychometric analysis	.54
	4.2.1.Validity	.55
	4.2.2.Realibility	.61
	4.3. Descriptives	.62
	4.4. Differential statistics	.65
	4.4.1. Correlations	.66
	4.4.2. Linear multiple regression	.67
	4.4.3. representativeness of the results	.69
5.	Discussion	.71
6.	Conclusion	.74
	6.1. Theoretical and managerial implications	.74
	6.2. Limitations and further research	.75
7.	References	.77
8.	Appendix	.90

LIST OF FIGURES

Figure 1. TAM	
Figure 2. UTAUT model	
Figure 3. The Health Belief Model	
Figure 4. Conceptual framework	40

LIST OF TABLES

Table 1. Participants country of origin	
Table 2. Measurements scales	49
Table 3. Total variance explained	57
Table 4. Rotated component matrix 1	58
Table 5. Rotated component matrix 2	59
Table 6. Rotated component matrix 3	60
Table 7. Item-total statistics	61
Table 8. Reliability statictics	62
Table 9. Mental health applications used	63
Table 10. Descriptive statistics of constructs	64
Table 11. Item correlations	67
Table 12. Linear multiple regression model summary	68
Table 13. ANOVA results	68
Table 14. Linear numerous regression coefficients	69
Table 15. Representativeness of the results	70
Table 16. Hypothesis overview and results	73

LIST OF ABBREVIATIONS AND ACRONYMS

ТАМ	Technology acceptance model
UTAUT	Unified theory of acceptance and use of technology
НВМ	Health belief model
PE	Performance expectancy
EE	Effort expectancy
SI	Social Influence
нм	Hedonic motivation
PSE	Perceived severity
PSU	Perceived susceptibility
SE	Self-efficacy
CA	Cues to action
BI	Behavioral intention
mHealth	Mobile Health
SPSS	Statistical Package for the Social Science
WHO	World Health Organization
NIMH	National Institute of Mental Health

1. INTRODUCTION

Mental illnesses such as anxiety, depression, social anxiety, or substance abuse are a growing problem in our society. Already, between 35% and 50% of people with mental illness do not receive treatment because suitable treatment places are scarce (WHO, 2012).

This each trend is leading to a growing range of solutions. Internet-based therapy programs are clinically effective in treating various mental disorders. One advantage of online therapy is its time and cost-efficiency. The amount of time clinicians spend with each client is much less than in regular face-to-face treatment, which means more clients can be served than in a traditional therapy setting (Forona, MacWilliams & McArthur, 2016).

1.1. RESEARCH PROBLEM

Despite compelling evidence of mental health apps' effectiveness and market growth (Marshall et al., 2020), the adoption of these new technologies is still relatively low. Mental health apps are not a routine part of mental health care, nor has a mental health platform been widely adopted by consumers (Bovin et al., 2019).

Having this in mind, it is essential to understand the factors that influence user adoption of mental health applications and derive suggestions for further improvement. Several theories and models have been presented to explain technologies' adoption and long-term use (Gupta et al., 2008). Although there is much research in technology acceptance and adoption, previous research suggests that different technologies have diverse factors that influence user behavior, meaning that the determinants that influence user acceptance of mental health apps may not be consistent with other technologies (Gupta et al., 2008).

More recent, unified models may explain more of the variance in adoption and use. Still, most of this literature has traditionally focused on adopting technology in a substantially different context than mental health treatment in the workplace (Connolly et al., 2018).

1.2. CONTEXT OF THE STUDY

Today smartphones play a central role in our lives. According to Statista, the current number of smartphone users worldwide is 6.378 billion, which means that 80.69% of the world's population owns a smartphone. This figure is significantly higher than in 2016, when there were only 3.668 billion users, corresponding to 49.40% of the world's population (Statista, 2021).

Due to the proliferation of smartphones and the Internet, the market for mental health apps has grown continuously in recent years and will continue to do so (MarketsandMarkets, 2020). Mental health apps target various mental disorders and vary in design and functionality. According to the NIMH (2017), they can be classified into six categories based on their functionality: Self-management, cognition enhancement, skills training, social support, symptom tracking, and passive data collection. Mental health apps span all stages of clinical care provision, including immediate crisis intervention, prevention, diagnosis, primary treatment, supplement to in-person therapy, and post-treatment condition management. Research suggests that mental health apps can positively influence a wide range of health conditions (Marcollino et al., 2018).

In addition, mobile health devices and apps have the potential to save resources, reduce the cost of care, increase outreach, and improve health outcomes (Ventola, 2014). Mental health apps offer even more significant potential, particularly for mental health conditions where stigma and lack of independence are additional barriers to seeking treatment (Bovin et al., 2019). According to a report by UnivDatos, there are currently over 300,000 health apps in mobile app stores worldwide, with the mental health segment representing the most significant growth market. It is forecasted to grow at a

CAGR of 20.5% from 2021 to 2027, reaching USD 3.3 billion by 2027. The COVID-19 pandemic has given a sudden boost to the digital market. The US's top 20 mental health apps reached 4 million first-time downloads in April 2020. That is up 29% from 3.1 million in January (UnivDatos, 2021).

Despite the positive outlook, there are differences in market growth. While North America dominates the mental health market, the share of Europe, Asia Pacific, and Latin America remain comparatively lower (MarketsandMarkets, 2020).

1.3. RESEARCH PURPOSE AND OBJECTIVES

The goal of this study is to analyze and understand the factors that influence the adoption of mental health apps. A better understanding of people's perspectives on mental health apps is needed to support the development of targeted implementation strategies and platform changes that ultimately promote adoption. This study aims to understand the extent to which performance expectancy, effort expectancy, social influence, perceived susceptibility, severity, action incentives, self-efficacy, hedonic motivation, and consumer intention influence user behavior.

1.4. RESEARCH SIGNIFICANCE

The increase in mental health problems, the impact of social media and COVID-19, along with a low capacity of therapy places, pushes to find alternatives for traditional help. Mental health apps offer the solution to this.

Growing advertising and increasing preference for mobile apps drive the health app market. Increasing smartphone penetration is underpinning the growth (Grand View Research, 2021). Nevertheless, it is essential to identify other factors that influence the adoption of mental health apps. The findings lead to implications for promoting and developing greater adoption of mental health apps. The success of

these apps depends on understanding people's concerns and identifying the factors that promote or inhibit their use.

1.5. RESEARCH QUESTION

In this case, this study aims to determine the acceptance of mental health applications among the population. To achieve this goal, this study targets to answer the following research question: Which factors influence the adoption of mental health applications?

It aims to identify and analyze the selected determinants in the conceptual framework: Performance expectancy, effort expectancy, social influence, perceived susceptibility, perceived severity, action incentives, self-efficacy, and hedonic motivation. In addition, the parameters will be analyzed in terms of participants' age, gender, and country of origin to identify demographic differences. Finally, the relationship between these factors and intention to use will be explored, i.e., whether there are negative, positive, or no ties and whether some elements have a more significant influence on consumer behavior than others.

1.6. OUTLINE OF THE REPORT

This work consists of six chapters. Each of them addresses a critical research topic.

Chapter 1 introduces the major themes of this research, focusing on the key factors that influence the adoption of mental health applications. A short introduction to the research field and a question statement were provided. The purpose and objectives of the research were explained in detail. In addition, the context of the study was presented in this chapter, followed by an explanation of the significance of the study. Finally, the thesis structure is outlined, and a brief description of each chapter is given.

Chapter 2 reviews and examines the existing literature and studies on mental health applications, highlights the research problem and identifies the significant factors that influence the adoption of mental health applications. This chapter presents mental disorders, mental health applications, and the theoretical frameworks used in this research area. A brief historical overview of mental health applications is provided, and the main advantages and disadvantages are examined. The primary models and theories developed and used to assess and describe individuals' acceptance and adoption of new technologies or products are introduced.

Chapter 3 addresses the research methodology. It identifies the research questions and explains the overall research design process and the justification of the research methods chosen. It begins by stating the research objectives and introducing the conceptual framework and the hypothesis developed. The respondents and the procedure by which the data were collected are described. This is followed by a presentation of the measures and materials and a description of the data collection process.

Chapter 4 presents the most important results of the study. It analyzes and discusses the quantitative results in light of the original conceptual model proposed in Chapter 3 about adopting mental health apps. The quantitative research was conducted through a questionnaire-based survey to explore the factors that influence the adoption of mental health apps. The chapter provides descriptive statistics of the data collected, discusses the implications of the survey results, and summarizes the main findings of the analysis. Finally, the proposed hypotheses are tested, and the conceptual model is revised based on the survey results.

Chapter 5 is the discussion part of the thesis. It contains the interpretation and explanation and will explain the implications of the results and make predictions and suggestions for future research.

Chapter 6 summarizes the empirical results with the literature and revises the proposed conceptual model based on the factors that have the most significant influence on the use of mental health apps. In addition, factors that emerged in this research and were not included in the original conceptual

model are considered. Finally, this chapter summarizes the study's findings, draws conclusions, highlights implications for research and practice, identifies limitations, and provides recommendations for future research.

2. LITERATURE REVIEW

2.1. OVERVIEW

This chapter reviews the relevant literature on the introduction of mental health applications. It begins with an introduction to mental disorders, a detailed discussion of mental health applications, and a presentation of various theories of technology acceptance and the Health Belief Model. Based on this, research gaps are identified, and hypotheses are developed.

2.2. MENTAL HEALTH DISORDERS

2.2.1. DEFINITION

According to the DSM-IV mental disorder is "a clinically significant behavioral or psychological syndrome or pattern that occurs in an individual. Associated with present distress (...) or with a significantly increased risk of suffering death, pain, disability, or an important loss of freedom (...)" (Stein et al., 2021). It encompasses our emotional, psychological, and social well-being and influences how we think, feel and act. However, there are differences between poor mental health and mental disorders. Even if a person is not diagnosed with a mental illness, they may still experience physical, psychological, and social unwellness (National Center for Chronic Disease Prevention and Health Promotion, Division of Population Health, 2021). Many different psychological issues are characterized by painful thoughts, perceptions, feelings, behaviors, and relationships with others. These include depression, bipolar disorder, schizophrenia, dementia, and eating disorders. Because the determinants of mental health and mental disorders are general, it is not only people with a genetic predisposition who become ill. Stress, diet, life experience, chronic illness, drugs, or loneliness can also lead to mental disorders (WHO, 2019).

2.2.2. EVOLUTION

Mental health is crucial for individual well-being and social and economic participation. In 2016, more than one in six people in EU countries had a mental health issue (OECD, 2018).

Mental disorders are more present than ever. Worldwide statistics about psychological health show that mental disorders have increased significantly over the past decade. These illnesses are one of the leading reasons for disability worldwide and the leading cause in the United States of America. Today, over 970 million people worldwide are affected by at least one of 200 forms of mental illness (SingleCare, 2021).

Anyone can suffer from mental health disorders, and they can impact a person's life as much as physical illnesses. The most commonly diagnosed mental conditions include depression, anxiety, and eating disorders. About a quarter of the population reported suffering from at least one mental illness (Stewart, 2021).

Thirty-four percent of Generation Z reported that their mental health deteriorated during the pandemic (American Psychological Association, 2020). An ongoing study funded by the National Science Foundation, NORC at the University of Chicago shows that younger people had more difficulty coping with the limitations resulting from the coronavirus and had higher levels of depression and anxiety (2020). Considering that quarantine has already affected people's behavior, a new phobia, corona phobia, emerged in 2020, with panic, anxiety, depression, paranoia, and obsessive-compulsive behavior (Brauser, 2020). Fifty-one percent of people in Europe reported that their mental health problems had worsened due to the COVID-19 pandemic. The age group most affected by poor mental health due to the pandemic is 18- to 24-year-olds (Stewart, 2021). Besides COVID-19, racism, prejudices, and bullying are still serious factors that impact people's mental health. Despite increasing attention to equality, the National School Climate Survey shows that LGBTQ+ adults are at increased risk for psychological health issues because of stigmatization and marginalization (2019).

Furthermore, there are differences in the number of people affected depending on gender. Women are at higher risk of experiencing psychological health disorders. Domestic violence, income, socioeconomic inequality, and social status are why more women than men suffer from depression and other psychological diseases (Statista, 2019). According to the latest global mental health statistics, about 9% of people suffer from disordered eating habits. Anorexia and bulimia are the two most frequent diseases (Deloitte Access Economics, 2020). Previous studies prove a correlation between eating disorders behavior and depression. Therefore, people affected by eating disorders are more likely to suffer from depression. Health statistics suggest that 36-50% of patients with bulimia nervosa also have a major depressive disorder (Levy , 2020).

Although people's mental health is a huge concern today; only 2% of the global budget is spent on its treatment, representing \$1 trillion per year. The World Economic Forum estimated \$16 trillion to treat all mental disorders by 2030 (World Economic Forum).

2.2.3. INFLUENCING FACTORS

Mental illness is becoming one of the most prevalent public health problems worldwide and a challenge in our society today. It is impacted by various determinants, including genetic predisposition, socioeconomic background, adverse childhood experiences, chronic illness, or substance abuse (European Commission, 2021). Young adults, called Gen Z, are affected by mental health disorders. In a study of current APA stress in America from 2019, more than nine in 10 Gen Z adults reported an experience in at least one physical or emotional symptom due to stress, such as feeling depressed or sad, having no interest, motivation, or energy. This is attributed to money, work, the political climate, and climate change (APA, 2019).

A study published in the Journal of Abnormal Psychology shows that depression rates among young adults increased significantly between 2009 and 2017. Based on the Centres for Disease Control and

Prevention database, suicide rates among 15- to 24-year-olds rose in 2017 to the highest level since 2000 (Miron et al., 2019). These high numbers can be explained by various factors, three of the most important factors which especially people nowadays have to deal with are stigmatization, COVID-19 and stigmatization (Gao et al., 2020).

Stigmatization

Negative attitudes and beliefs toward people who have mental illness are widespread. Mental disorders lead to more negative judgments and stigma than any other illness. Stigma can cause discrimination, such as negative comments about mental illness or treatment. Many patients bear from social exclusion, prejudice, and experience bullying, physical violence, or harassment (Rössler, 2016). InIn conclusion, individuals with psychological disorders are deprived of the opportunities that make up a good life: good jobs, safe housing, and adequate health care (Corrigan & Watson, 2002).

Even though the middle ages are long gone, the general population is ignorant about mental health disorders, and fear of the mentally ill remains widespread (Rössler, 2016). Public stigma is the general population's reaction to people with mental diseases. Self-stigma is the prejudgement that individuals with psychological disorders direct against themselves. Stereotypes, prejudice, and discrimination drive public stigma and self-stigma. Stereotypes, in particular, has a significant impact because most members learn them of a social group (Hilton, 1996; Watson et al., 2006). The public stigma attached to mental illness is pervasive. There is no country or society where people with mental illnesses have the same social status as people without mental disorders (Rössler, 2016). In a survey involving 27 different countries, nearly 50% of people with schizophrenia reported discrimination. Up to ½ of those affected by this psychological disease anticipated discrimination when applying for jobs (Thornicroft et al., 2009). Behavioral effects resulting from public stigma are evident in refusal of help, avoidance, coercive treatment, and segregated facilities (Weiner et al., 1988).

It is hazardous when people with mental illness internalize the stigma. Self-stigma lowers self-esteem and self-efficacy, limiting the prospects for recovery. This can usually happen before or when people are not affected by mental disorders. People typically learn and internalize culturally prevalent stereotypes without even getting into contact with such conditions (Rössler, 2016).

Social Media

Another stressor people have to deal with is the immense use of social media. Social media has nowadays become essential in people's daily activities. Many individuals spend hours on platforms like Instagram, Facebook, and TikTok (Bartosik-Purgat et al., 2017).

Social media are interactive online and mobile networks through which people and communities share, create, or spread information, pictures, or short clips on a platform. Social networks are an essential form of communication in many people's lives and continue to grow (Berger et al., 2005). The use of applications such as Instagram, YouTube, and TikTok has become enormously popular for social interactions. They have become much more important than traditional media, especially in young adults' lives (Michikyan & Suárez-Orozco, 2016).

Social contact with others can alleviate stress, anxiety, and depression facilitated through social media. In today's world, many of us rely on social media platforms like Facebook, Twitter, Snapchat, YouTube, and Instagram to socialize and connect. Even if each of these platforms has an advantage, it's important to remember that social media can never be a substitute for genuine human relationships. Face-to-face contact with other human beings is necessary to trigger the hormones which causes happiness and positivity and decrease stress and anxiety. Ironically, technology that brings people closer together can make you feel even lonelier and more isolated by spending too much time online - and exacerbate mental health issues like anxiety and depression (Hunt et al., 2018). Since Social media is a relatively new technology, there is little research about the long-term consequences. However, concerns about the impact of intensive use of these media are rising. In particular, the effect on young people's mental health can be significant (American Association for Suicidology, 2017). Some studies have already shown evidence of a link between social media use and mental health issues (Karim et al., 2020) and stated that these technologies are responsible for aggravating mental health disorders (Karim et al., 2020). A previous study on the impact of social media on mental health and well-being shows that young people perceive social media as a threat to mental well-being. It is believed to cause mood and anxiety disorders, is seen as a platform for cyberbullying, and the use itself has often been described as addictive (O'Reilly et al., 2018).

About 10 percent of teens report being bullied on social media, and many other users are subjected to abusive comments. Social networks can be hotspots for spreading hurtful rumors, lies, and namecalling that can leave lasting emotional scars. The given anonymity through the internet reduces the barriers to disrespectful interaction with others (Hasebrink et al., 2011). Several studies have shown robust associations between cyberbullying and mental health, particularly suicide and self-harm (Daine et al., 2013). Social media is often seen as a fake world. Even if people are aware of manipulated and heavily edited photos, they can still convey a sense of insecurity about one's appearance or life (Hunt et al., 2018).

Online Networks lead to constant and harmful comparison with the lives of others, which studies have identified as a risk factor for decreased well-being. A variety of studies show the link between social media use and negative self-esteem and self-image (Education Policy Institute, 2021). In particular, the idea of an idealized body image impacts self-esteem, especially among young women, with 9 out of 10 teenage girls reporting that they are dissatisfied with their bodies (Karslidou & Thomas, 2021). Manipulated images on social media platforms lead to unrealistic expectations for young people. Online advice and information risks trivializing and normalizing unhealthy behaviors and can lead to conditions such as eating disorders (Bell, 2007).

The constant comparison with other individuals and other individuals' lives can also lead to the fear of missing out. This phenomenon has been around for much longer, but social media's significance impacted it. According to Przybylski et al. (2013), FOMO is a pervasive fear that others may have rewarding experiences from which one is absent. It is characterized by a desire to stay connected to what others are doing. As people share photos of their supposed highlights, others may think they are missing out on certain things. This can affect an individual's self-esteem or cause anxiety. FOMO can lead to heavy social media use, restless nights, or dangerous accidents when people look at their cell phones even while driving (Karslidou & Thomas, 2021). The fear can lead not only to intensive use of social media but also to addiction. Social Media are intended to be addictive, and using them activates the brain's reward system by releasing dopamine. About 5% of adolescent users are considered more addicted to social media than alcohol and cigarette use (Jenner, 2015).

According to the OECD, extreme Internet users have lower overall life satisfaction than moderate users. Some studies have found that introverted young adults develop an addiction to further feedback on social media. This addiction can lead to poor sleep patterns or impaired performance on exams (OECD, 2016). Especially in the last two years since the COVID-19 pandemic outbreak, the use of social media has increased, and therefore its impact can be observed. A 2020 study conducted in China showed a high prevalence of mental health problems positively associated with frequent social media use during the COVID-19 outbreak (Gao et al., 2020). Another study at the University of Pennsylvania confirmed the link between feelings of loneliness and social media use. Therefore, reducing Facebook, Snapchat and Instagram can make people feel less lonely and isolated and improve their overall wellbeing (Hunt et al., 2018).

COVID-19

Another stressor people have to deal with today is COVID-19. Since the coronavirus in 2019, the global community has been concerned about the long-term physical and economic impact and the

psychological impact. The newly identified coronavirus, first reported in Wuhan, China, spread rapidly across the globe, causing numerous infections and deaths, especially among the elderly and vulnerable (Wang et al., 2020). Although efforts to control and limit the spread of the pandemic in the community are relatively straightforward to follow, prejudices and fears appear to be jeopardizing response efforts (Ren et al., 2020).

Currently, governments worldwide are focused on testing, treating infected individuals, developing drugs, vaccines, and treatment protocols. Although one might think that the pandemic is under control after almost two years, this is not the case. And most importantly, the long-term consequences and impact on people's mental health cannot be predicted (WHO, 2020d). Public health emergencies can affect individuals' and communities' health, safety, and welfare (Pfefferbaum & North, 2020). People tend to feel anxious and uncertain when the environment changes. When there are outbreaks of infectious diseases where the cause or course of the disease and the consequences are unclear, rumors and a narrow-minded attitude develop (Ren et al., 2020).

From previous studies on the SARS outbreak in 2003, about 70% of people in Hong Kong were afraid of contracting the virus and felt more at risk, unlike other diseases (Cheng & Cheung, 2005). This fear and anxiety about infection can lead to discriminatory actions. Accordingly, at the beginning of the COVID pandemic in 2019, discriminatory behavior towards Asians was evident. People from Wuhan have been targeted and blamed for the virus outbreak using terms such as "Wuhan virus" or "China virus" (Ren et al., 2020). Necessary measures such as self-isolation and quarantine have interfered with usual activities and routines, increasing people's loneliness, anxiety, and depression (WHO, 2020c). The amount of negative news, confrontation with death, and fear of Covid have negatively impacted the psyche of many individuals. Fear is a well-known response to infection outbreaks, and people react to the perceived threat differently. It can translate into a range of emotional reactions, unhealthy behaviors, and noncompliance with public health guidelines among sufferers and the general population (Pfefferbaum & North, 2020). In addition, millions of people have lost their jobs and family members. Over one hundred thousand people have died worldwide, and the numbers continue to rise (WHO, 2020b). Individuals, families, and communities experience feelings of hopelessness, despair, grief, loss, and profound loss of meaning due to pandemics (Levin, 2019). The feeling of losing control leads to fear and uncertainty about the course of the pandemic (Zikmund-Fisher et al., 2018). A study from 2020 conducted by the Indian Psychiatric Society shows a twenty percent increase in mental illness since the coronavirus outbreak in the country (Loiwal, 2020). Another research of 1210 respondents from 194 cities in China from January to February 2020 found that 54% rated the psychological impact of the COVID-19 outbreak as moderate or severe, 29% reported moderate to severe anxiety symptoms, and 17% reported depressive symptoms (Wang et al., 2020).

After disasters, most people are resilient and do not become psychopathological. However, depressive disorders and anxiety are significant concerns. Some groups are more vulnerable to the psychological effects of the pandemic than others. In particular, people with COVID, individuals at increased risk, and humans with pre-existing medical, psychiatric, or addiction problems are more vulnerable to adverse psychosocial effects. The burden on health care providers should not be underestimated either. The frequent contact with covid patients, overtime, and psychological stress place significant demands on them (Pfefferbaum & North, 2020). Also, the life of young adults and children changed a lot. Young people who are particularly socially active lack exchange with other individuals. Experts already predict that Gen Z will be more mentally affected by the pandemic than different generations (Glazer, 2020).

Generation Z is arguably the most socially attuned generation of all time (Chillakuri, 2020). Still, the ease of keeping up with everything going on in the world via the internet also has side effects. An APA study (2018) on American stressors found that news is a regular source of stress for young adults. Young adults are more likely than older people to be diagnosed with an anxiety disorder and report being diagnosed with depression (Goldman, 2020). People's reaction to fear and intolerance of uncertainty leads to negative social behaviors aimed at reducing uncontrolled situations that people fear (El-Terk, 2020). Domestic violence has increased worldwide because victims have no way to escape perpetrators through incarceration (Abramson, 2020). It is essential to provide psychosocial support and help gris evidently, the Covid pandemic will leave damage. Still, it is not estimable how significant the impact will be on the population's mental health (Li et al., 2020). Therefore, mental health and treatment options will play a more significant role in the future (Levin, 2019).

2.3. MENTAL HEALTH APPLICATIONS

Since mental health applications are a relatively new technological invention, it is necessary to explain. The term "mental health applications" should be defined comprehensively to understand how it originated and developed. Due to the positive market outlook regarding these technologies, it is essential to identify the determinants and influencing factors that may contribute to the acceptance and thus the adoption of these applications, keeping in mind that these factors may vary in different regions and societies.

2.3.1. DEFINITION

The term "mental health application" refers to tools that can be accessed via smartphone or mobile devices that focus on improving different aspects of mental health and well-being. These apps focus on treating or preventing various diseases by offering multiple features like journaling, meditation, or mood-tracking. They can also provide self-help tools or therapeutic activities to improve mental health (Morin, 2021). Today, several providers such as Moodfit, Calm, Happify, or Headspace differ in their functions and disorder focus. The individual's mental health should be improved by regularly using emotion-based activities such as breathing exercises, behavioral, social or thought-based activities (Morin, 2021). Mental health is the foundation for an individual's well-being and effective functioning. It is not just the absence of mental disorders but the ability to think, learn and understand one's feelings and the reactions of others. Mental health means balancing physical, psychological, social,

cultural, spiritual, and other interrelated factors (WHO, 2017). Disruption of these factors can lead to severe disorders, generally characterized by abnormal thoughts, perceptions, feelings, behaviors, and relationships. The most frequent psychological disorders are depression, bipolar disorder, schizophrenia, dementia, and anxiety. Today, there are effective therapies or medical treatments to treat these disorders and ways to prevent them. However, access to medical care and social services is essential to address these issues (WHO, 2019).

2.3.2. HISTORY

Because of the broad use of smartphones and the creation of Apple's iTunes App Store and Google's Android Market that allows users to download mobile applications in 2008 (Statista, 2021), more and more mental health applications are being developed and adopted (Clay, 2021). Since the early days of mental health apps, the market has grown steadily. According to the American Psychological Association, App stores host nearly 20,000 different mental health apps today, ranging from AI chatbots and mood detectors to services like Talkspace and BetterHelp that match patients with licensed therapists (Pappas, 2020). Forecasts show that the market size of Health apps will generally increase at a compound annual growth rate (CAGR) of 17.7% from 2021 to 2028 (Grand View Research, 2021).

One of the first mental health apps available was Headspace. It was officially founded in 2010 as an event company by Andy Puddicombe, who wanted to teach others about meditation and mindfulness. Wanting to make his techniques available to anyone, anytime, anywhere, Andy and a small team around him developed the idea of the app. Today, Headspace offers guided meditations, animations, articles, and videos that bring meditation and mindfulness closer to improving health and enhancing well-being in this world. With millions of users in more than 190 countries, the company seeks to achieve its goals (Headspace, 2021).

The shift from traditional models of care to patient-centered models is expected to increase the adoption of such applications. Growing smartphone penetration, use of the Internet and social media, and the number of healthcare professionals recommending mental health apps further influence the market growth (Grand View Research, 2021). Significantly since 2020, the demand for mental health apps has increased dramatically due to the Covid-19 pandemic. According to the findings of a survey, half of Americans have experienced negative mental health impacts from the pandemic. First-time downloads of the top 20 mental health wellness apps in the U.S. increased 29 percent from January to April, reaching 4 million. Compared to previous years, these apps typically peak in January due to New Year's resolutions while decreasing in the following months, which was not the case during the pandemic (Kirzinger et al., 2020). For example, Talkspace, which offers video- and text-based therapies, saw nearly double the number of users between mid-March and early May 2020 compared with the same period in 2019 (Levy, 2020). Talkspace competitor BetterHelp also sees an unprecedented surge in demand. According to CEO Alon Matas, the number of users opting for the platform specifically to help with stress and anxiety has doubled. According to Sensor Tower, downloads increased from 50,000 in January to 80,000 in April 2020, at least in the U.S. (Herzog, 2020).

Looking at downloads and user behavior by region, differences in adoption can be seen. These variations in market share may be due to cultural differences, access to devices, interest, information, or range of applications (Lipschitz et al., 2019). In 2020, North America dominated the market with more than 38% revenue share. Forecasts show that the regional market will continue to grow at a steady CAGR from 2021 to 2028 due to smartphone usage, development of care networks, increasing prevalence of chronic diseases, and rising geriatric population (Grand View Research, 2021). Besides the current market share, the Asian region is expected to exhibit the fastest CAGR over the forecast period, justified by the high demand for connected devices. Technological advancements and increased product demand are expected to boost new companies' market entry. Moreover, strategic promotions by companies in the form of mergers and acquisitions are likely to fuel competition and drive market growth.

However, there are also challenges that companies have to face. Concerns like skepticism and mistrust regarding data privacy and security may inhibit the adoption of mobile health technology and are factors hindering market growth (Grand View Research, 2021).

2.3.3. MARKET OUTLOOK

According to Research and Markets (2021), the global mental health application market will likely grow at a CAGR of around 31% in 2021-2026. This is due to multiple factors, including a growing awareness of health applications among various industry players and increasing initiatives for collaborations by multiple governments and market players. Many players lead multiple mindfulness and meditation applications for a new global wellbeing collaboration. These initiatives and partnerships are expected to provide mental health exercises, guided meditations, and sleep content to millions of employees, corporate clients, and guests and will be made available globally in the coming years.

Although the COVID-19 outbreak hurt almost all industries worldwide, the pandemic offers immense opportunities for the growth of the mental health app market. Today, iOS is the largest Appstore and is expected to capture the largest market share in the forecast years. The Appstore contains more health apps than its competitors and adds about 100 new health apps every day. Globally, North America has dominated the mental health app market in recent years and is likely to continue to do so. Due to the growing awareness of mental health issues and the abundant availability of numerous apps, the regional market is expanding (Research and Markets, 2021).

2.3.4. BENEFITS

Mental health apps target various mental disorders and vary in design and functionality. They fall into six categories: self-management, cognition improvement, skills training, social support, symptom tracking, and passive data collection (NIMH, 2017). Mental health apps cover all phases of clinical care,

including immediate crisis intervention, prevention, diagnosis, initial treatment, adjunct to personal therapy, and post-treatment condition management (Price et al., 2014).

Experts believe that these new technologies have great potential for both clients and doctors. These mobile apps are convenient because treatment can occur anytime, anywhere and can be ideal for those with difficulty with in-person appointments. Users can feel more anonymous and have access to treatment throughout the day. In addition, the cost can be lower than traditional treatment, and the services can be made available to more people (NIMH, 2017). By offering effective options to patients with milder psychiatric symptoms, the burden on traditional mental health services could be reduced. In general, cognitive apps can prevent illness or as an adjunct to conventional therapy (Newman et al., 2011).

2.3.5. DRAWBACKS

Despite the positive outlook for mental health apps, there are still some challenges that providers need to address. Despite their increasing use, the adoption of mental health apps still has room for improvement. One reason is the deficit of awareness and knowledge about these apps. People either don't know they exist, or they don't know how to find the right app for their needs (Lipschitz et al., 2019). It is often complicated and overwhelming for users to choose the right app from hundreds of apps available on the app market (Bashir, 2017).

Although there is evidence about the effectiveness of smartphone-based mental illness treatment, there is still mistrust in these new technologies (Marshall et al., 2020). There are no industry-wide standards for consumers to determine whether an app or other mobile technology has proven valid. Further research shows that the most commonly cited reasons for not using mental health apps are weak evidence of effectiveness. These findings suggest that public dissemination of information about the validation of mental health apps could improve adoption (Lipschitz et al., 2019). Another issue is the lack of personalization. It's essential to understand whether mental health apps work for everyone and under conditions. Even though the choices are vast, it is often difficult to get an overview and choose the app that best meets personal needs (NIMH, 2017). Privacy and data autonomy concerns have also weighed on the teletherapy field. Psychological health apps deal with sensitive private information and therefore need to ensure user privacy. Mental health sensitivity can be attributed to the long-standing social stigma that prevents people from seeking mental health treatment and support (Corrigan, 2004). Previous research shows that stigma can also be an issue when using mental health applications. Some people don't want others to know that they are using a mental health app, therefore these apps should be designed discrete and be password protected (Kenny et al., 2014).

Furthermore, negative publicity increases distrust in these apps. In February 2020, Jezebel reported that BetterHelp and Talkspace shared data with third parties. This data included anonymized intake forms with sensitive information about users' mental health, sexual orientation, and suicidal thoughts (Osberg, 2020). Sharing data with third parties is pervasive. In 2019, researchers examined the data practices of 36 leading depression and smoking cessation apps and showed that more than 80 percent of the apps transferred data to Facebook and Google, often without disclosing it in their privacy policies (Huckvale et al., 2019). When developers send user data to Mix Panel or Facebook, they can aggregate and commercialize it without their knowledge. The dangerous matter is that this behavior is not even prohibited, and there is no way to prevent the data from being shared with a fourth party. More transparency and regulation must be created to create more trust in this digital therapy space (Herzog, 2020). There are significant concerns about whether these apps can adequately protect mental health information. Informing users about how data is protected within the app can increase uptake (Lipschitz et al., 2019).

Because mental health applications deal with a sensitive topic, they differ in their acceptability from other types of technologies. For this reason, traditional technology acceptance models may not be as accurate in their application.

2.4. THEORETICAL FRAMEWORKS

2.4.1. TECHNOLOGY ACCEPTANCE MODEL

Several influential factors stand out in reviewing the current literature on mental health app adoption. Most studies analyzing the introduction, adoption, and long-term use of mental health apps are based on the technology acceptance model which short form is TAM, the most widely used model of user acceptance, and user behavior (Holtz et al., 2020).

The TAM stems from the theory of reasoned action, which intends to describe the relationship between dispositions and behaviors within people's actions. It is mainly used to predict how individuals behave based on their pre-existing attitudes and behavioral intentions (Taherdoost, 2017). TAM, in turn, is an information technology framework for understanding users' adoption and use of new technologies and assumes that perceived usefulness and perceived utility predict technology acceptance (Ma & Liu, 2005). Therefore, consumers are more likely to adopt new technology if it has a high-quality user experience design, and they expect to benefit from its use (Portz et al., 2019).

Figure 1

Technology acceptance model



Note: Adopted from Davis, 1985.

Since its invention, the TAM has been tested with various applications and is still the most common model to predict the acceptance of new technologies. Nevertheless, some results have statistical significance, direction, and magnitude weaknesses. Even though the correlation between usefulness and acceptability and between use and usability is strong, the relationship between usability and acceptability is weak, and its significance does not pass the fail-safe test (Ma & Liu, 2005). About mental health apps, the predictive power of TAM is also being studied. However, different studies show contradictory results, considering that they were conducted at different moment in time and countries. Research conducted in Germany in 2016 about the acceptance of mobile apps for mental health treatment shows no significant direct influence of perceived usefulness on behavioral intention regarding mental health apps. The indirect effect is present, as perceived usefulness partly reflects the population's knowledge of these apps, their effectiveness, and their ability to treat illness. While some participants indicate that mental health apps provide helpful information, most do not believe these technologies are sufficient for treating mental disorders. Regarding the usability of mental health apps, the study shows that this factor does not significantly influence usage behavior because the young German participants are experienced with smartphone use and do not consider mental health apps a challenge. Nevertheless, the study shows a positive correlation between perceived usefulness and ease of use (Becker, 2016). Another analysis focuses more on beliefs about the continued use of mental health apps after implementation confirms the direct effect of perceived ease of use on perceived usefulness. However, it states that ease of use significantly predicts satisfaction with mental health apps. Both effectiveness and ease of use significantly and positively influence intention to continue using them (Cho, 2016).

Another study shows that perceived ease of use and usefulness influence the acceptance of mental support apps. However, this study also indicates that the effect is relatively small compared to the power of subjective norms like internal motivation or subjective importance of the topic. And thus, this is not the only study that shows that the TAM is not sufficient to predict consumers' usage behavior regarding mental health apps (Lim & Wong, 2019).

2.4.2. EXTENDED TECHNOLOGY ACCEPTANCE MODEL

Since its invention, the TAM has evolved as results show that additional and more significant factors influence technology adoption. In general, demographic characteristics have been studied along with other influences. Previous research has confirmed that gender, country of origin, and age can significantly affect mental health app use (Cho, 2016).

In general, the society in which people are socialized has a significant impact on their mental health awareness. People aware of mental health are generally more likely to seek treatment. However, most people do not feel the force for mental health app usage because they consider themselves healthy and not at risk of psychological disorders (Becker, 2016).

An analysis of the technology acceptance model about e-learning reveals additional major influencing factors. The research interest focused on studying the acceptance of e-learning courses and what factors influence the adoption of such approaches. The results show that self-efficacy is the factor that most influences behavioral intention, followed by social impact (Park, 2009).

Depending on the technology and the nature of the study, the basic TAM framework was extended differently. Hsu and Chang (2013) attempted to study Moodle adoption using an extended TAM. To do

so, they added the external variable perceived convenience. The results show that perceived convenience influences perceived usefulness and intention to use Moodle. Other researchers tested the TAM on other applications. Malhotra et al. considered trust as another influencing factor. Therefore, a lack of trust in the tool data security influences people's usage behavior (Malhotra et al., 2004). When using mental health apps, sensitive data about the client is collected, which can lead to privacy concerns. Researchers confirm a correlation between trust and usage behavior when using mental health apps (Becker, 2016).

2.4.3. UNIFIED THEORY OF ACCEPTANCE AND USE OF TECHNOLOGY

The unified theory of acceptance and use of technology, in short UTAUT, is a model formulated by Venkatesh et al. in 2003. It aims to explain users' intentions to use an information system and subsequent usage behavior. The theory consists of four central constructs. According to them, performance expectancy, effort expectancy, social influence, and facilitating conditions influence user acceptance of technology (Venkatesh et al., 2003). In addition, the variables gender, age, experience, and voluntariness of use indirectly influence usage behavior (Abbad, 2021).

Figure 2

The unified theory of acceptance and use of technology



Note: Adopted from Venkatesh et al., 2003.

Performance expectancy refers to how people believe using the system will help them. Therefore, a high-performance expectancy increases the intention to use a particular technology. Effort expectancy refers to the degree of ease associated with using the system. Individuals who expect the system to be easy to use are more likely to use it. The third factor, social influence, represents individual perceptions of others' importance on using the technology. People who know someone who uses a specific technology recommended it or took part in a conversation about it are more likely to adopt it (Abbad, 2021).

The perceived extent to which the organizational and technical infrastructure required to support the technologies is referred to as facilitating conditions (Thomas et al., 2013). Studies have consistently shown that infrastructural and organizational aspects are essential variables to consider when implementing new information systems. The possession of a smartphone, which is necessary for using mental health apps, is part of the factor and the knowledge needed to use them or an environment that can help you (Phichitchaisopa & Naenna, 2013). Although the UTAUT has already been applied based on some technologies to predict their adoption, there are still research gaps when it comes to validity regarding the adoption of mental health apps in Europe. Yueh et al. investigated the factors

influencing students' use of wikis using the UTAUT. The results show that effort expectancy and social influence have the most significant impact on use by students (2015).

Another study that focused on user experiences with mobile health applications for patients with eating disorders found that health care providers and health experts were more likely to report barriers to mobile health application adoption than facilitators, suggesting that mHealth technologies are challenging to obtain the use. Most factors influencing the adoption of mobile health applications were attributed to external factors related to the environment, such as time, rather than internal factors related to individual barriers. Nevertheless, participants reported adoption barriers such as the inability to personalize the app a lack of motivational or interactive components (Anastasiadou et al., 2019).

2.4.4. UNIFIED THEORY OF ACCEPTANCE AND USE OF TECHNOLOGY 2

Nine years after the UTAUT invention, Venkatesh et al. published an extended version of the Unified theory of acceptance and use of technology. It additionally includes factors relevant to the consumer market that influence behavioral intention to use new technologies. Venkatesh et al. extracted the usage intention factor from the original UTAUT and expanded it to include three elements to improve the prediction of behavioral intention and usage behavior (2012).

The extended model additionally includes hedonic motivation, defined as the pleasure or enjoyment derived from using a technology; price value, which represents consumers' cognitive trade-off between the perceived benefits and monetary costs of operating a technology (Dodds et al., 1991); experience and habit, defined as the extent to which people tend to perform behaviors automatically based on learning. Combined with the preceding factors, they should predict people's behavioral intention (Limayem et al., 2007).

Even though the UTAUT2 is one of the most widely used technology acceptance models, studies still suggest an adjustment. Using this model, a survey of 317 participants showed that performance expectancy, hedonic motivation, and habit positively predicted users' intention to use a health and fitness app, whereas effort expectancy, social influence, facilitating conditions, and the price was entirely but not significantly associated with intent to use (Yuan et al., 2015).

2.4.5. HEALTH BELIEF MODEL

Regarding the adoption of disease treatments and prevention, other literature focuses on the health belief model, in short HBM. The HBM was invented in the early 1950s by social scientists at the U.S. Public Health Service to understand why people do not adopt disease prevention strategies or screening tests for early detection of disease. It is one of the most widely used theories in health behavior and is used as both an explanatory theory and a theory of change (Janz & Becker, 1984). The HBM assumes that a person's belief in a personal threat of disease, together with a trust in the effectiveness of the recommended health behavior or intervention, predicts the likelihood that the person will adopt the behavior (LaMorte, 2019).

The HBM is based on psychological and behavioral theories. A value-expectancy approach compares advantages and disadvantages or risks and benefits for decision-making (Franklin Health research Foundation, 2021). The basis of the model consists of two components: the desire to prevent or cure a disease and the belief that a particular action will prevent or cure a disease. In addition, the person's behavior often depends on the perception of the benefits and barriers associated with the interventions (LaMorte, 2019).

The theory includes six constructs that relate to how a person decides whether or not to engage in a particular behavior. The main factors that make up the model are perceived susceptibility, perceived severity, perceived benefits, perceived barriers, cues to action, and self-efficacy (Becker, 1974).

Perceived susceptibility refers to a person's subjective perception of their risk of contracting a disease. The sense of openness to condition varies widely. Due to cognitive biases, the general public cannot assess susceptibility. For example, people generally overestimate the risks of conventional drugs while underestimating the dangers of natural products (LaMorte, 2019).

Perceived severity asks whether the outcome in question is essential or not. It indicates how serious the individual perceives the infection with a disease and includes the imposition of both the corporal and social results (Franklin Health research Foundation, 2021). An individual's perception of the effectiveness of various interventions to reduce the risk of disease is part of the perceived benefit factor. Even if individuals believe they are at serious risk of illness, they will not engage in the treatment or prevention if not perceived to be effective (LaMorte, 2019).

Perceived barriers represent a person's feelings about the obstacles to taking a recommended health action and lead to a cost-benefit analysis. Barriers to the activity or unshaven, including financial costs and time, social costs, physical discomfort, and other emotional costs. Stimuli for action are the stimuli needed to trigger the decision process. These stimuli can be internal, such as pain or constraint, or external, like advice from friends or advertising.

The last factor, self-efficacy, was defined by Bandura and referred to the belief that someone can perform a specific behavior (Franklin Health research Foundation, 2021). The Health belief model assumes that messages produce optimal behavior change when they successfully target perceived obstacles, benefits, self-efficacy, and threats. Although the model appears to be an ideal explanatory framework for communication research, theoretical limitations have limited its use in this field (Jones et al., 2016). Researchers have argued that the HBM does not specify the order of variables, which is essential for researchers interested in understanding communication processes (Champion & Skinner, 2008). It does not make habitual or perform behaviors for non-health-related reasons, such as social acceptance. It counterfeits that all individuals have the same access to the disease or condition information. Therefore, social influence and facilitating conditions should also be influencing factors.
The HBM is descriptive rather than explanatory and does not suggest a strategy for changing healthrelated actions. The individual constructs are helpful depending on the health outcome of interest. Still, for the most effective use of the model, it should be integrated with other models that consider the environmental context and suggest strategies for change (LaMorte, 2019).

Figure 3

The health belief model



Note: Adopted from Janz & Becker, 1984.

According to the acceptance of psychological health applications, the HBM can be a significant contribution to predicting the behavioral intention to use this new technology. It can complement classic technology acceptance models because it does not only focus on the adoption of new

technologies but covers the health-related part. Therefore, combining both models could forecast and explain health app adoption more accurately.

3. METHODOLOGY

3.1. OVERVIEW

This chapter establishes the conceptual framework and research method for the empirical study. The methodological framework guides for choosing research methods and consists of philosophical assumptions, logical reasoning, and criteria for evaluating research findings (Scotland, 2012). Based on positivism, deductive approach, and quantitative method, a questionnaire survey is designed in this chapter. Measurement scales are designed based on previous empirical studies; data collection and analysis procedures are discussed; data quality is assessed; ethical issues are comprehensively addressed.

3.2. RESEARCH OBJECTIVES

The previous academic world has focused on the technology adoption model and the U.S. market to adopt mental health applications. However, some factors that influence end-user behavior have not yet been considered, especially in the European market. With combining the unified theory of acceptance and use of technology (Venkatesh et al., 2003) and the Health belief model(1950) as a foundation, this study contributes to research on mental health apps by providing a new approach to the topic. The results will give the academic world closer insights into the factors influencing consumers' intention to use mental health apps in the European market and may serve as a basis for further or long-term research.

3.3. CONCEPTUAL FRAMEWORK AND RESEARCH HYPOTHESIS

Building on the existing literature, a combination of the UTAUT 2 (Venkatesh et al., 2012) and the HBM (Janz & Becker, 1984) was chosen to serve as the basis for the conceptual framework study. The Unified

Theory of Technology Acceptance and Use has comparatively better explanatory power than other technology acceptance models. It is best suited for examining usage behavior from the consumer context rather than the organizational context (Woldeyohannes & Ngwenyama, 2017). Previous studies have demonstrated its usefulness in predicting usage intentions and adopting new technologies (Abrahao et al., 2016).

Because mental health applications are a form of health technology, it is essential to incorporate factors included in the Health belief model into the conceptual framework of this study to analyze the particular case of mental health app adoption more precisely (Abraham & Sheeran, 2015). The conceptual framework presented below (Figure 4) was conducted based on secondary research in the technology adoption field. The model includes eight factors that directly influence the behavioral intention that leads to the user behavior of the consumers and three indirect determinants, such as gender, age, and country of origin.

Figure 4

Conceptual framework



Unlike the original models UTAUT2 and HBM, the conceptual framework differs in some aspects. In contrast to the UTAUT2, the factors facilitating conditions, price value, habit, and experience were omitted (Venkatesh et al., 2012). In this study, reducing needs would primarily be represented by the ownership of smartphones. However, since it is assumed that every participant owns a smartphone or would have the opportunity to own one, the analysis of this factor would not be meaningful. This research focuses primarily on non-users of mental health apps, as they are considered a potential market. Because of this, it is essential to analyze the factors that influence their behavioral intentions to adopt these technologies. Accordingly, the elements of habit and experience can be omitted, as both can just be answered by respondents who are using a mental health app consistently. Also, the factor price value is not discussed further in this study since it results from an individual cost-benefit

analysis, which is not applicable. It is also impossible to make a generalized statement about the costs and benefits of mental health apps, as they vary widely in both features and prices.

Furthermore, the factors perceived susceptibility, perceived severity, self-efficacy, and cues to action that originated from the HBM are included in the conceptual framework. The left determinants are perceived benefits and barriers, which have already been covered by combining other determinants from the UTAUT2. In addition, perceived benefits and barriers are very general factors and would therefore need to be considered more in detail. The analyzed in this study will be discussed in more detail formulate the hypotheses in the next step.

Behavioral Intention

Behavioral intention is defined as a customer's intention to adopt and use a particular tool (Venkatesh et al., 2003). According to Irani et al. (2009), most research on technology adoption has used behavioral intention to predict user behavior and, therefore, technology adoption. Behavior in this study refers to whether or not a participant intends to adopt a mental health app.

Performance expectancy

According to UTAUT, performance expectancy refers to how users assume that using new technology will improve their effectiveness in performing specific tasks or benefit them and influences whether or not a person will adopt the behavior. These expectations are influenced by the person's gender and age (Venkatesh et al., 2003). Sun et al. (2013) suggested that effectiveness is primarily reflected in how they help users reduce health-related threats. Therefore, positive performance expectations may increase intention to use the technology, given common concerns about health-related threats.

H1: Performance expectancy positively influences behavioral intention.

Effort expectancy

Generally, effort expectancy has widely viewed how users perceive that adopting new technology will be free of effort (Gao et al., 2015). It plays a vital role in using a particular technology and is influenced by gender, age, and other experience. Dabholkar (1996) cited two main reasons to adopt technology: reducing effort and social risk. Featherman and Hajli (2016) describe social risk as consumers' belief that they will look foolish in front of others. This can be the case when consumers believe that technology is too difficult to use (Venkatesh et al., 2003), and they would not use it at all then ask for help. A more accessible application to use than another is more likely to be adopted (Pikkarainen et al., 2004). Effort expectancy refers to the ease of use and perceptions of the time required to select and use a mental health app (Venkatesh et al., 2003). This thesis means the ease of use of mental health apps.

H2: Effort expectancy negatively influences behavioral intention.

Social influence

The social influence factor reflects how individuals' decision-making is influenced by social factors such as subjective norms and the expectations of significant others (Venkatesh & Morris, 2000). Many studies based on the UTAUT model have demonstrated the importance of social influence in technology adoption and implementation (Alraja et al., 2016). Dwivedi et al. (2016) found that peer evaluation exerts normative pressure on users when mental health app adoption is visible to others. Furthermore, decision-making processes are always influenced by the personal reference group and tend to meet their expectations (Bearden & Etzel, 1982). Cultural differences may also affect this. People living in a more collectivistic society such as China may feel more social pressure and subjective norms than individuals in an inductivist society where greater importance is placed on achieving personal goals and self-image (Hofstede & Bond, 1984). The role of social influence factors in decisionmaking-making has been demonstrated in empirical studies (Scholz et al., 2020). Therefore, it can be hypothesized that people with positive social influence have a higher intention to use mental health applications.

H3: Social influence positively influences behavioral intention.

Perceived susceptibility

As mentioned earlier, the conceptual framework of this study is based not only on the UTAUT but also on the Health Belief Model. Therefore, perceived susceptibility is mentioned as another determinant. It is about individuals' perceived risk of developing a health impairment. The HBM predicts that individuals who perceive themselves as vulnerable to a particular health problem will participate in behaviors that decrease their health problem risk (Adhikari, 2019). In terms of mental health apps, this would imply that people who see themselves at risk of developing one are more likely to use them.

H4: Perceived susceptibility positively influences behavioral intention.

Hedonic motivation

Hedonic motivation is the enjoyment or pleasure derived from using new technology (Venkatesh et al., 2012). These factors play an essential role in technology adoption and use (Brown & Venkatesh, 2005). The decision-making process is not only controlled by cognition but also by emotions. Therefore, positive emotions positively influence the intention to use new technology (Ha & Stoel, 2009). In terms of mobile health service adoption, studies have shown that hedonic motivation varies across countries. Dwivedi et al. (2016) reported a positive influence of Bangladeshi citizens' intention to use mobile health services, while the effect was insignificant in Canada and the United States. Therefore, the role of hedonic motivation needs to be explored in more detail. This factor was included in the conceptual framework to analyze and test its predictive power for intention to use mental health apps.

H5: Hedonic motivation positively influences behavioral intention.

Perceived severity

Perceived severity, also called perceived seriousness, means the subjective assessment of the severity of a health problem and its potential consequences. HBM assumes that individuals who perceive a particular health problem to be serious are more likely to engage in behaviors that prevent the health problem from occurring (Hochbaum, 1958). Perceived seriousness includes both beliefs about the illness itself and its disease in functioning at work and in social roles. Even if a person does not perceive the disease as medically serious, they may perceive the financial or social consequences as serious. These results may relate to an anticipated happening in the future or a current condition, such as a pre-existing health problem (Rosenstock, 1974). A previous study of Pap smears for cervical cancer screening showed that the mean severity score among persons who underwent Pap smears was higher than the mean perceived severity score among persons who did not take the test. Perceived severity is higher among those who have experienced a Pap smear, indicating that those who fear the negative consequences of the disease are more likely to seek behavioral changes to prevent it (Abotchie & Shokar, 2009). This suggests that it is similar to the use of mental health apps.

H6: Perceived severity positively influences behavioral intention.

Self-efficacy

Another hypothesis is based on self-efficacy. As part of HBM, it represents self-confidence in one's ability to implement and act on a behavior. Some studies question Self-efficacy because research

shows that its effects are often related to whether a person performs the desired behavior (Adhikari, 2019).

H8: Self-efficacy positively influences behavioral intention.

Cues to action

The last factor is influencing the adoption of disease prevention or treatment strategies are the cues to action. This is the incentive required to trigger the decision-making process to adopt a recommended health intervention. These incentives can be internal or external (Adhikari, 2019). Therefore, people who have previously heard about, been recommended, or seen advertisements for mental health apps are more likely to use them.

H7: Cues to action positively influence behavioral intention.

3.4. PARTICIPANTS

For this study, a sample of 309 individuals was surveyed between Nov. 28, 2021, and Dec. 31, 2021. Participants were between 18 and 70 years old, with a mean age of 32. 71.2% of respondents are between 21 and 36 years old, with an exceptionally high percentage of 21- to 29-year-olds. figureThe participants in this study are 57% female, 41.7% male, and 1.3% non-binary or third gender. In terms of country of origin, more than half of the respondents were from Portugal or Germany, as shown in table 1. The United States and Spain followed them. The other countries were only represented by less than 10 participants per country, representing 24.6% of the survey participants.

Table 1

Country	Frequency	Percentage	Cumulative percentage
Germany	78	25.2	25.2
Portugal	83	26.9	52.1
Spain	14	4.5	56.6
United States of America	58	18.8	75.4
Others	76	24.6	100
Total	309	100.0	

3.5. PROCEDURE

The data collection was conducted using a self-administered online questionnaire because it allows standardized and easily comparable data to be collected from a large population quickly, inexpensively, and in a very economical manner (Ponchio et al., 2021).

It is an appropriate strategy to measure the variables included in the hypotheses on a larger scale. The use of a questionnaire survey has the following advantages for this study. It maximizes the objectivity of the data and can be generalized to a larger population. The researchers are separated from the subjects during data collection, and the questions are highly structured and standardized, eliminating influence and confounding factors (Fricker & Schonlau, 2002). A standardized and structured format allows researchers to analyse the data more efficiently, and the use of statistical tools further reduces researcher interference (Melkert & Vos, 2010). Numerical data also provides higher accuracy, comparability of values, and graphical analysis. In addition, a questionnaire survey allows researchers to reach a large number of subjects simultaneously and collect a large sample at a relatively low cost

(Collis & Hussey, 2013). Because participants must be familiar with the Internet and social media, online questionnaires are appropriate for reaching this target audience.

However, it should be noted that an online survey also has some disadvantages. First, the standardized design offers less flexibility and less in-depth information. Respondents are not free to express their true thoughts and are limited to predetermined answers (Couper, 2008). In addition, respondents may not take the questionnaire seriously or may be inaccurate, affecting the results' reliability (Hoonakker & Carayon, 2009).

The questionnaire was developed on Qualtrics, an online platform for web-based surveys, and distributed through a link on several social networks, namely Facebook, Instagram, and LinkedIn, and individually via messaging platforms such as Messenger and WhatsApp. Confidentiality of the results was guaranteed, and it was ensured that the data collected were for purely academic purposes and would be analyzed in compliance with the GDPR. To provide the proper conduct of the research, the questionnaire was submitted, reviewed, and approved by the Research Ethics Committee of Nova IMS before it was distributed.

3.6. MEASURES

3.6.1. QUESTIONNAIRE DESIGN

This study used a questionnaire consisting of four main groups: (i) sociodemographic questions, (ii) experience with mental health applications, (iii) influencing factors, (iv) behavioral intention.

The first group was dedicated to the socio-demographic questions of the respondents, namely age, gender, and country of origin.

The second part, composed of questions about experience with mental health apps, asked respondents whether they had ever used a mental health app or were currently using a mental health app. If so, whether they could name the mental health app they were using.

The third group includes questions about the eight factors that may influence the adoption of mental health apps. Therefore, it consists of eight blocks of items. In the first block, participants answer four questions about the performance expectation of mental health apps. This is followed by questions on effort expectancy and social influence. The fourth and fifth blocks asked about hedonic motivation. They perceived susceptibility, representing perceived susceptibility to mental disorders—the following sections address perceived severity and self-efficacy, followed by the final block on cues to action.

The fourth block was dedicated to respondents' behavioral intention. It aimed to clarify whether participants intended or planned to use a mental health app.

3.6.2. MEASUREMENT SCALES

Because this study was conducted worldwide, all questionnaire questions are written in English (Appendix 1).

For this study and to standardize the measurement scales, a five-point Likert scale was used for most questions, ranging from "strongly disagree" (1) to "strongly agree" (5), to rate all main items according to the participants' level of agreement with the statements. Five-point Likert scales, which indicate a smaller range of opinions, are appropriate for measuring participants' attitudes. In addition, there is a disadvantage to selecting more options. Studies show that attention span reaches its effectiveness with six votes at a time. When assessing possible mood levels, our minds can only consider six at a time, so respondents have to invest a lot of time remembering possible choices. Five-point Likert scales have been shown to ensure that respondents do not lose interest. Analyzing the different response options is also easier for respondents because they are not tempted to select middle options or leave

boxes blank. In addition, five sentiment levels provide a reasonable range of opinions that allow for a comprehensive understanding of attitudes toward the object or phenomenon being assessed (Taherdoost, 2020).

According to the overview of previous empirical studies, all the variables included in the conceptual framework of this study have been covered by previous studies. Hence, the measurements used in these studies can be borrowed by this thesis after slight modifications.

UTAUT 2 questionnaire

The research design was guided by Venkatesh et al.'s (2012) seven original constructs of technology acceptance. Sixteen items were adapted from the existing literature, with all questions relating to mental health apps. These items cover five of the nine constructs Performance Expectancy, Effort Expectancy, Social Influence, Hedonic Motivation, and Behavioral Intention.

Health belief model questionnaire

To measure perceived susceptibility, perceived severity, self-efficacy, and cues to action, 15 items were adopted from the existing literature to assess the four constructs. These items are based on the health belief model questionnaires, which emerged from U.S public health researchers Godfrey Hochbaum, Stephen Kegels, Howard Leventhal, and Irwin Rosenstock. Later the first precise formulation of the HBM appeared in a paper by Rosenstock in 1966 and was later refined by Marshall Becker (1974).

Table 2

Measurement scales

Constructs	Number	Items	Sources
	of items		
Performance expectancy	4	Using a mental health app would improve my mental wellbeing.	Venkatesh et al. (2012)
		Using a mental health app can satisfy my mental health care needs.	
		Using a mental health app can improve my efficacy in monitoring my mental health conditions.	
		Using a mental health app will improve the quality of my life.	
Effort expectancy	4	Choosing a mental health app would require a lot of effort.	Venkatesh et al. (2012)
		Using a mental health app would require a lot of effort.	
		Using a mental health app would require a lot of time.	
		Learning how to use a mental health app would be difficult.	
Social influence	3	People have asked me about my mental health.	Venkatesh et al. (2012)
		People have already advised me to seek mental health help.	

		People have already advised me to use a mental health app.	
Hedonic motivation	3	I think using a mental health app is fun.	Venkatesh et al. (2012)
		I think using a mental health app is entertaining,	
		l think using a mental health app is enjoyable.	
Behavioral intention	4	l intend to use a mental health app.	Venkatesh et al. (2012)
		I intend to check the availability of a suited mental health app.	
		l plan to use a mental health app.	
		It is worth using a mental health app.	
Perceived susceptibility	4	Everybody can get a mental health disease.	Becker (1974)
		l am not at risk of mental health disease.	
		I can have a mental health disease even without feeling its signs and symptoms.	
		I am afraid of getting a mental health disease.	
Perceived severity	4	Mental health diseases can lead to death.	Becker (1974)
		Mental health diseases can change the whole life.	

		Mental health diseases can disrupt the harmony in families.	
		Mental health diseases are long lasting.	
Self-efficacy	3	I can always manage to solve difficult problem if I try hard enough.	Becker (1974)
		It is easy for me to accomplish my goals.	
		I am confident that I could deal efficiently with unexpected events.	
Cues to action	4	l heard about mental health apps before.	Becker (1974)
		I saw an advertisement about a mental health app.	
		I know at least one person who is using a mental health app.	
		Someone has already recommended a mental health app to me.	

3.7. DATA COLLECTION

This study used non-probabilistic random sampling based on the relatively easy availability of participants and is commonly used when the population to be studied too large to include. Not all individuals in the people have an equal chance of participating in this study, which influences the representativeness of the study. Still, it is the most cost-effective and least time-consuming method (Mweshi & Sakyi, 2020).

A power analysis was conducted using G-Power 3.1.9.7 software to determine the sample size required for this study. The results indicated a minimum sample size of 251 participants to achieve a statistical power of 99% for a mean effect size of 0.08 at a significance level of 1% (0.01) for the proposed model (Verma & Verma, 2020). Still, a larger sample was used to minimize the error further. In addition, Kyriazos (2018) recommends a ratio of 5 to 10 participants per item for a minimal example of 100 respondents, suggesting a selection of 195 to 390 participants for this study.

Before analyzing the results, the data were preprocessed and cleaned by identifying and deleting incomplete, irrelevant, and incorrect responses. A total of 438 participants completed the questionnaire. However, when the data was cleaned, it was found that 29.5% of the responses were incomplete and therefore removed from the data set, leaving a total of 309 completed questionnaires. Further analysis revealed no anomalies or errors. Therefore, the data set with 309 samples were used for the calculations.

4. **RESULTS**

4.1. OVERVIEW

For data analysis, several statistical tests were performed using IBM SPSS - Statistical Package for the Social Sciences (version 27). First, the analysis of the psychometric properties of the instruments is performed to ensure the validity and reliability of the studied constructs. Second, descriptive and differential statistics are presented to analyze the means and standard deviations of the constructs. Sociodemographic characteristics such as gender, age, and country of origin are compared concerning the use of mental health applications and the constructs mentioned in the conceptual framework. In addition, correlation analysis was performed to determine the associations and intensity between each construct. Linear multiple regression tests each hypothesis and assesses the influence of the independent variables on the dependent variables. The conceptual framework can be tested as a predictive model, and all eight ideas can be verified or falsified as appropriate. Statements can also be made about the strength of correlations. The data analysis is concluded with a goodness-of-fit test to examine whether the results can be generalized to the population.

4.2. PSYCHOMETRIC ANALYSIS

The psychometric analysis is the structured process of measuring the psychometric properties of an indicator by analyzing test data and ensuring the quality of the survey used. Because the accuracy of the results depends on the quality of the questionnaire, it is essential to analyze the psychometric properties, including the validity and reliability of the items used to measure the constructs (Jones & Thissen, 2006).

Validity explains how well the data collected cover the actual domain of investigation (Ghauri & Gronhaug, 2005). It means that an item measures what it was intended to measure (Field, 2005). There

are different subtypes of validity, such as criterion validity, face validity, content validity, and construct validity. In this case, only construct validity is used. It refers to how well a construct has been transformed into an operationalization (Taherdoost, 2016).

On the other hand, reliability concerns how a measurement of a phenomenon yields stable and consistent results (Carmines & Zeller, 1979). It is concerned with repeatability, which means that a repeated samplesize produces the same results (Moser & Kalton, 1989). Reliability is measured by Cronbach's alpha coefficient, which is considered good at 0.70 or higher (Hair et al., 2018).

4.2.1. VALIDITY

The validity of the nine constructs was tested using principal component analysis (PCA). This variable reduction technique is used when a construct is measured by more than one item. PCA can therefore be used to check whether all items measure the same build and whether they can be transformed into a new variable with or without excluding one thing. (PCA)

For conducting a PCA, sample size recommendations are given by many authors, according to which the sample size of 309 participants collected for this study is considered suitable for factor analysis (Kyriazos, 2018). For this PCA, the Kaiser-Meyer-Olkin indicator (KMO indicator) and Bartlett's test of sphericity were used to assess whether the correlations between items were acceptable, which is the case for KMA values above .70 and Bartlett's test at a statistical significance of p < .05. Component extraction was based on Kaise-Guttmann criteria, scree plot analysis, and the percentage of variance explained, which is considered suitable for values above 60.0% (Hair et al., 2018). Only items with a factor-item correlation greater than .40 were selected in this analysis.

UTAUT2 questionnaire

A PCA was conducted to assess the validity of the items measuring performance expectancy, effort expectancy, hedonic motivation, and social influence. The KMO indicator (.81) and Bartlett's test for sphericity [χ 2 (91) = 1793.65, p < .001] were used to demonstrate adequate measurement of the constructs.

The eigenvalue of a factor indicates how much of the total variance of all variables is explained by that factor. The so-called "Kaiser criterion" (also "eigenvalue rule") states that only factors whose eigenvalue is more significant than 1.0 should be extracted. SPSS selects the number of factors strictly according to this criterion. As shown in Table 3, four elements have eigenvalues greater than 1.0. The middle column "Cumulated %" indicates that these four factors together explain 68.73% of the variance of all variables.

Table 3

Total variance explained

				Extr	action sum	s of squared
		Initial eige	nvalues		loadir	ngs
		% of			% of	
Component	Total	variance	Cumulative %	Total	variance	Cumulative %
1	4.21	30.12	30.12	4.22	30.12	30.12
2	2.60	18.59	48.71	2.60	18.59	47.11
3	1.74	12.39	61.10	1.74	12.39	61.10
4	1.07	7.63	68.73	1.07	7.63	68.73
5	.85	6.03	74.76			
6	.61	4.38	79.14			
7	.49	3.49	82.63			
8	.47	3.36	85.99			
9	.42	2.97	88.96			
10	.38	2.73	91.70			
11	.33	2.36	94.05			
12	.32	2.26	96.32			
13	.28	1.99	98.31			
14	.24	1.69	100.00			

Extraction method: Principal component analysis.

The following table shows the rotated factor loadings of the data which represent both, the weighting for each factor and the correlation between the factors. A factor loading on a variable is the correlation between the variable and the element. For example, variable PE_1 and component 1 correlate at .805. The magnitude of the factor loading indicates how closely a variable is related to an element: amounts close to 0 show little correlation. The higher the value, the closer the correlation. Each variable is assigned to the factor it loads most strongly. Accordingly, it can be confirmed that the first four items measure the construct 'performance expectancy, the items HM_1, HM_2, and HM_3 can be assigned to the factor hedonic motivation, the construct effort expectancy is represented by EE_1 to EE_4, and the fourth-factor social influence is measured by the items SI_1, SI_2 and SI_3.

Table 4

Rotated component Matrix 1

	Component					
	1	2	3	4		
PE_4	.805					
PE_3	.797					
PE_2	.730					
PE_1	.693					
HM_2		.864				
HM_1		.842				
HM_3		.811				
EE_3			.838			
EE_2			.820			
EE_1			.751			
EE_4			.686			
SI_2				.865		
SI_1				.850		
SI_3				.697		

Extraction method: Principal component analysis. Rotation method: Varimax with Kaiser normalization.

a. Rotation converged in 5 iterations.

HBM questionnaire

Considering the values of the KMO indicator (.82) and Bartlett's test for sphericity [χ 2 (171) = 2680.12, p < .001] obtained by PCA, we can demonstrate the adequate measurement of five factors explaining 66.50% of the variance.

Nevertheless, the rotated component matrix (table 5) shows that item PSU_1 was assigned to the first

factor, perceived severity, although it should be given to the construct perceived susceptibility. For this

reason, it is not included in further analyses. The item PSU_2 is also excluded in the other course since it only shows a weak correlation to one of the factors.

Table 5

Rotated component matrix 2

	Component				
	1	2	3	4	5
PSE_2	.882				
PSE_3	.872				
PSU_1	.791				
PSE_1	.783				
PSE_4	.570				
BI_1		.914			
BI_3		.908			
BI_2		.904			
BI_4		.759			
CA_3			.795		
CA_2			.774		
CA_4			.709		
CA_1			.680		
SE_3				.815	
SE_2				.748	
SE_1				.722	
PSU_2	.329				
PSU_4					.821
PSU_3					.587

Extraction method: Principal component analysis.

Rotation method: Varimax with Kaiser normalization.

a. Rotation converged in 5 iterations.

After exclusion of PSU_1 and PSU_2, PCA was performed repeatedly. The results are shown in Table 6. The KMO indicator (.80) and Bartlett's test for sphericity [χ 2 (136) = 2372.14, p < .001] prove the adequate measurement of five factors explaining 68.98% of the variance. The first factor is composed of the items related to behavioral intention. The second component relates to perceived severity and is measured by PSE_1 through PSE_4. CA_1, CA_2, CA_3, and CA_4, the third factor, is calculated, representing action cues. Self-efficacy is the fourth factor measured by three additional items. The last construct, perceived susceptibility, is calculated by PSU_3 and PSU_4.

Table 6

Rotated component matrix 3

	Component					
	1	2	3	4	5	
BI_1	.917					
BI_3	.913					
BI_2	.908					
BI_4	.752					
PSE_3		.894				
PSE_2		.883				
PSE_1		.801				
PSE_4		.637				
CA_3			.795			
CA_2			.778			
CA_4			.706			
CA_1			.685			
SE_3				.834		
SE_2				.766		
SE_1				.755		
PSU_4					.873	
PSU_3					.615	

Extraction method: Principal component analysis.

Rotation method: Varimax with Kaiser normalization.

a. Rotation converged in 5 iterations.

4.2.2. RELIABILITY

An essential measure of internal consistency and thus of reliability of a questionnaire is Cronbach's alpha. It indicates the extent to which the individual questions of a questionnaire agree with each other and measure the same construct. Thus, when internal consistency is high, the answers to the individual questions roughly concur. In other words, the question items all correlate positively with each other. A Cronbach's alpha above 0.7 is considered sufficient (Hair et al., 2018).

The Cronbach's alpha for the four items measuring the construct performance expectancy is .82 and for effort expectancy .78. When calculating the construct social influence (.75), one can see in table 7 that one can increase Cronbach's alpha.78 if one deletes the last item. Therefore, the third item is excluded in further analysis.

Table 7

Item-total statistics

				Cronbach's
	Scale means if	Scale variance if	Corrected item-	alpha if item
	Item deleted.	item deleted	total correlation	deleted
SI_1	4.93	5.778	.599	.644
SI_2	5.45	5.423	.671	.555
SI_3	5.87	6.797	.476	.778

The three items measuring hedonic motivation have a reliability of .88, as shown in table 8. The construct perceived susceptibility, whose articles also showed some irregularities in the factor analysis, offers weak reliability of .25 when all items are included. If the first item is excluded, the reliability increases to .53, which is still low and must be considered critically in further analysis.

The construct perceived severity is measured with four items. If the fourth item is excluded, the reliability increases from .84 to .86. The calculation of Cronbach's alpha for self-efficacy results in a value of .71 for all three things together and .76 for the cues to action. The construct behavioral intention shows the highest internal consistency with a value of .92. If the fourth item is omitted, the reliability increases to .94.

Table 8

Reliability statistics

Constructs	Cronbach's alpha	N of items
Performance expectancy	.82	4
Effort expectancy	.78	4
Social influence	.78	2
Hedonic motivation	.88	3
Perceived susceptibility	.53	3
Perceived severity	.86	3
Self-efficacy	.71	3
Cues to action	.76	4
Behavioral intention	.94	3

4.3. DESCRIPTIVES

After evaluating the psychometric properties of the measurement instruments, the analysis of the descriptive statistics of the variables was performed as follows. To better understand the characterization of the sample about the topic of mental health apps, several descriptive analyses were conducted.

First, about participants' experience with mental health apps, it was found that nearly 50% of participants had ever used a mental health app. Still, only 17.8% continued to use them, meaning that 54 participants in the study used a mental health app for data collection. In total, 20 different mental

health apps were mentioned as used by participants. The most commonly used mental health apps are Calm and Headspace, followed by other apps that you can see in table 9.

Table 9

Mental health applications used by participants

		Frequency
Name		
	Calm	13
	Headspace	12
	Happify	3
	7Mind	3
	Sanvello	3
	Talkspace	2
	Good fit	2
	InnerHour	2
	Betterhelp	2
	GuidanceResources	2
	Balloon	1
	Breathe	1
	HealthPlix	1
	Insight Timer	1
	MindBeacon	1
	Peak	1
	Sesh	1
	Smiling Mind	1
	Woebot	1
	Wysa	1
	-	

25.6% of German participants currently use a mental health app, and 4.82% Portuguese participants. Of the participants from the USA, as many as 34.48% use a mental health app. In terms of gender, 18.75% of female participants and 17.05% of male participants currently use a mental health app. 71.2% of the respondents are between 21 and 36 years old, 211 people. 18.18% of them were using a mental health app during data collection. And 17.72% of those over 36 years old also reported using a mental health app.

Then, the descriptive analysis was conducted for the primary constructs studied. It was considered that a five-point rating scale from "strongly disagree" (1) to "strongly agree" (5) and a three-point rating scale from "no" (1) and "maybe (2) to "yes" (3) were used to measure the main questionnaire items, and that the results of each construct were calculated by adding the values assigned to each item; the higher the average value, the higher the agreement with the influencing factors. Respondents' perceptions of the influencing factors and behavioral intentions are shown below (Table 10).

Table 10

	Number of		
	point	Mean	Std. deviation
PE	5	3.53	.77
EE	5	2.85	.88
SI	5	2.93	1.30
HM	5	3.24	.93
PSU	5	3.23	.73
PSE	5	4.43	.77
SE	5	3.50	.80
CA	3	2.14	.67
BI	5	3.18	1.11

The data show that participants have a relatively high-performance expectation of mental health apps, meaning that they believe in the effectiveness of these apps and see them as a good source of help. In terms of effort, expectations are relatively mixed below. Also, the social influence of using a mental health app does not seem to be very pronounced among participants; instead, hedonic motivation is slightly higher. It indicates how participants expect utilizing an application to be fun. Survey participants also tend to rate themselves as more prone to mental disorders. The most apparent value is that for perceived severity, which shows that mental illness is taken seriously. The value for the construct call to action also indicates that the topic of mental health is present. The behavioral intention is also slightly above average, implying increased use of mental health apps in the future.

4.4. DIFFERENTIAL STATISTICS

A comparative analysis of the results was then performed according to the socio-demographic characteristics of the respondents, such as gender, age, and country of origin. Gender group analysis revealed no significant difference between groups (p > .05), except for effort expectancy. (Female (M = 2.74, SD = .87, n = 176) participants had, on average, lower effort expectancy than males (M = 2.98, SD = .85, n = 129). According to Cohen (1992), the effect size is r = .28, corresponding to a weak effect.

In terms of age, participants were divided into two groups depending on the average age of the respondents. Group one consisted of those 18 to 32 years old. Group two included those aged 33 to 70. These two groups showed significant differences in social influence (t(307) = 3.42, p = .001), perceived severity (t(307) = 44.44, p < .001), and calls to action (t(307) = 3.10, p = .002). The Cohen (1992) effect size for social influence is r = .40, corresponding to a medium effect. For perceived severity, it is r = .53, which corresponds to a strong effect. For cues to action, it is r = .37, which means that the younger group perceives mental illness as more severe, was more often asked about their mental health by their environment, and already had more contact with the topic of mental health apps.

The last mean comparison is between German and Portuguese respondents due to the high participation. Looking at the data, some significant differences emerge. First, the experience of using a mental health app is significantly higher among German participants (M = 1.55, SD = .501, n = 78) than among Portuguese participants (M = 1.30, SD = .46, n = 83); second, more German respondents

currently use a mental health app. The two groups also differ significantly in performance expectancy (t(159) = 2.27, p = .024), calls to action (t(159) = 3.83, p = < .001), and behavioral intention (t(159) = 2.56, p = .012). This means that German participants believe more in the effectiveness of mental health apps, that the topic is more present in their social environment, and that they are more inclined to use one of these apps.

4.4.1. CORRELATIONS

Subsequently, the correlations between the variables were analyzed using the Pearson correlation coefficient. A significant moderate positive correlation was demonstrated between the construct hedonic motivation and performance expectancy (r = .589, p < .001) (r between .30 and .70). Similarly, between the variables hedonic motivation and perceived receptivity (r = .315, p < .001), performance expectancy and self-efficacy (r = .310, p < .001), hedonic motivation and self-efficacy (r = .323, p < .001 and perceived receptivity and self-efficacy (r = .248, p < .001). Furthermore, a correlation is found between behavioral intention and performance expectancy (r = .619, p < .001), hedonic motivation (r = .618, p < .001), and cues to action (r = .363, p < .001). As shown in table 11, the higher the performance expectancy, hedonic motivation, and cues to action, the higher the behavioral intention to use a mental health app.

Table 11

	PE	EE	SI	HM	PSU	PSE	SE	BI
PE								
EE	017							
SI	.138*	.153**						
НМ	.589**	.021	.075					
PSU	.295**	.174**	.075	.315**				
PSE	.111	033	026	003	.174**			
SE	.310**	.064	091	.323**	.248**	.025		
BI	.619**	059	.207**	.618**	.181**	022	.169**	
CA	.287**	104	.194**	.270**	.117*	.060	.129*	.363**

*. Correlation is significant at the 0.05 level (2-tailed).

**. Correlation is significant at the 0.01 level (2-tailed).

4.4.2. LINEAR MULTIPLE REGRESSION

Considering the previous results, multiple regression was performed using the enter method to examine the effects of each factor on the intention to use a mental health application. Regression indicates the directional linear relationship between two or more variables. The so-called coefficient of determination (R²) expresses how well the regression line reflects the relationship between the independent and dependent variables and should be between zero and one (table 12). The dependent variable is behavioral intention, the predictors are performance expectancy, effort expectancy, social influence, hedonic motivation, perceived susceptibility, perceived severity, self-efficacy and cues to action.

Table 12

Linear multiple regression model summary

			Adjusted R	Std. The error of
Model	R	R Square	Square	the Estimate
1	.727 ^a	.529	.517	.7684

a. Predictors: (constant), CA, PSE, EE, SE, SI, PSU, HM, PE

The determination of a regression function does not yet mean that the determined correlation is significant. The so-called F-test determines the significance of the regression. The results show that the linear model is statistically substantial t [F(8, 300) = 42.15, p < .001] and that the set of variables explains 24.89% of the variance in behavioral intention (Table 13).

Table 13

ANOVA results

		Sum of				
Mode	l	squares	df	Mean square	F	Sig.
1	Regression	199.114	8	24.889	42.150	.000 ^b
	Residual	177.149	300	.590		
	Total	376.263	308			

a. Dependent variable: BI

b. Predictors: (constant), CA, PSE, EE, SE, SI, PSU, HM, PE

Additionally, the influence of each factor within the model was analyzed. It was found that within this combined group of factors, performance expectancy (t = 7.306, p = <.001), social influence (t = 2.426, p = .016), hedonic motivation (t = 7.517, p = <.001), and action incentives (t = 3.328, p = <.001) were

the only constructs with a significant influence on behavioral intention (Table 14). Therefore, H1, H3, H4, and H8 could be verified.

Table 14

Linear multiple regression coefficients

		Unstandardized	Coefficients	Standardized		
Model		B	Std. Error	Beta	t	Sig.
1	(Constant)	.112	.389		.289	.773
	PE	.540	.074	.375	7.306	.000
	EE	066	.052	052	-1.257	.210
	SI	.086	.035	.102	2.426	.016
	HM	.461	.061	.385	7.517	.000
	PSU	058	.066	039	876	.382
	PSE	089	.059	062	-1.512	.132
	SE	092	.060	066	-1.522	.129
	CA	.240	.072	.143	3.328	.001

a. Dependent variable: BI

The variables effort expectancy, perceived susceptibility, perceived severity, and self-efficacy have no significant influence on behavioral intention. This means that they alone are not sufficient to predict consumer behavior. However, variables such as perceived susceptibility and self-efficacy can still correlate with behavioral intention. This only indicates the degree of relationship between the variables, which in this case is weak. For this reason, H2, H5, H6, and H7 cannot be verified.

4.4.3. REPRESENTATIVENESS OF THE RESULTS

In addition, a goodness-of-fit test was performed to determine whether the sample data corresponds to the data that would be expected in the actual population. It indicates the discrepancy between the observed values and the values that would be expected from the model in a normal distribution. Representativeness is defined as drawing accurate conclusions about a population from a sample. The results of the chi-square goodness-of-fit test show a significant impact for each variable (Table 15). This means a substantial difference between the observed data and the expected values.

Table 15

Representativeness of the results

	PE	EE	SI	HM	PSU	PSE	SE	BI	CA
Chi-Square	243.087	126.343	49.107 ^b	194.634	216.087	656.709	205.405	103.761	50.505 ^b
	а	а		С	d	d	С	С	
df	16	16	8	12	11	11	12	12	8
Asymp. Sig.	.000	.000	.000	.000	.000	.000	.000	.000	.000
a. 0 cells (0,0%) have expected frequencies less than 5. The minimum expected cell frequency is 18,2.									

b. 0 cells (0,0%) have expected frequencies less than 5. The minimum expected cell frequency is 34,3.

c. 0 cells (0,0%) have expected frequencies less than 5. The minimum expected cell frequency is 23,8.

d. 0 cells (0,0%) have expected frequencies less than 5. The minimum expected cell frequency is 25,8.

5. DISCUSSION

Over the years, numerous studies have been conducted on mental health apps, mainly focusing on efficacy and privacy (Marshall et al., 2019; Parker et al., 2019). However, mental health apps are a relatively new technology and are only now being explored.

Given the rise in mental disorders and the lack of treatments, there is a growing recognition of the need to explore mental health apps and their use as an alternative to traditional therapy (NIH, 2021). In this field, it is essential to understand the characteristics that most influence the adoption of these technologies to meet consumer needs and improve uptake. Considering that mental health is a sensitive issue in our society, the factors mentioned in classical models such as the Technology Acceptance Model or the Unified Acceptance Theory may not be sufficient to explain the acceptance of mental health applications. Therefore, other characteristics related to the health belief model were additionally proposed and investigated.

Accordingly, the main objective of this study was to understand which of the following factors such as performance expectancy, effort expectancy, social influence, hedonic motivation, perceived susceptibility, perceived severity, self-efficacy, and cues to action influence behavioral intention to use a mental health application. The results on the research questions of the proposal confirmed that performance expectancy, social influence, hedonic motivation, and cues to action positively influence the behavioral intention to use mental health applications. The effect of effort expectancy, perceived susceptibility, perceived severity, and self-efficacy on behavioral intention was not significantly demonstrated.

First, the study results show that almost 50% of the participants have used a mental health app before, but only 17.8% continue to use it. This suggests that participants are open to using these apps but are not satisfied with them. There are also significant differences between users in different countries, such as Germany and Portugal. The study shows that there is no significant difference in use in terms
of the age of the respondents. On average, participants have high-performance expectations, believe in mental health apps' effectiveness, and have positive attitudes toward these technologies. Their perceived severity and susceptibility are relatively high, indicating that they are aware of mental disorders and are more willing to use a mental health app in the future than not to use one. Strikingly, on average, respondents were not exposed to many cues of action, such as people who have recommended a mental health app to them, who know people who use such an app, or who have seen an advertisement about it.

Correlation analyses showed a significant relationship between performance expectancy, social influence, hedonic motivation, perceived susceptibility, self-efficacy, cues to action, and behavioral intention. The existence of a correlation is not sufficient to make a statement about the predictive power of the items. Therefore, multiple regression was conducted to test the hypothesis based on the conceptual framework. The first hypothesis aimed to investigate the influence of performance expectancy on behavioral intention based on the UTAUT 2 (Venkatesh et al. 2003), which was supported by the obtained results that showed a significant influence of performance expectancy on behavioral intention (H1). These results are comparable to a study about Hospital Patients' adoption of medical apps (Chang et al., 2021).

It was then hypothesized that effort expectancy negatively influences behavioral intention (H2). Different from the UTAUT 2 model (Venkatesh et al. 2003), this study could not find a significant effect of this factor on behavioral intention. Nevertheless, social influence was shown to positively impact on behavioral intention (H3). Thus, when participants' acquaintances ask them about their mental health, advise them to seek help, or recommend using a mental health app, this positively influences their intention. The fourth hypothesis, indicating a positive influence of hedonic motivation on behavioral intention, was also confirmed (H4). Accordingly, participants who believe that using a mental health app is fun, entertaining, and enjoyable are more likely to use such an app. In contrast to the Health belief model(Becker, 1974), data from this study failed to show a significant influence of perceived

susceptibility and perceived severity on behavioral intention (H5 & H6). This means that even if people feel vulnerable or take mental disorders seriously, it does not influence their behavioral choice to use a mental health app. Also, the positive influence of self-efficacy on behavioral intention could not be demonstrated in this study (H7), unlike in other studies regarding help-seeking intentions reported (Langley et al., 2017). The last hypothesis could be verified (H8). According to this hypothesis, cues to action positively influence behavioral intention. This means that participants who have heard of apps, seen an advertisement for an app or know people who use and have recommended a mental health app are more likely to use an app.

Table 16

Hvp	othesis	overview	and	resul	ts
-----	---------	----------	-----	-------	----

Hypothesis	Relationship	Results
H1	Performance expectancy -> behavioral intention	Supported
H2	Effort expectancy -> behavioral intention	Rejected
H3	Social influence -> behavioral intention	Supported
H4	Hedonic motivation -> behavioral intention	Supported
H5	Perceived susceptibility -> behavioral intention	Rejected
H6	Perceived severity -> behavioral intention	Rejected
H7	Self-efficacy -> behavioral intention	Rejected
H8	Cues to action -> behavioral intention	Supported

6. CONCLUSION

This study aimed to examine the influence of performance expectancy, effort expectancy, social impact, hedonistic motivation, perceived susceptibility, perceived severity, self-efficacy, and calls to action on behavioral intention to use a mental health application.

The study followed a hypothetico-deductive quantitative approach to test the research hypothesis, and data were collected using a self-administered online questionnaire consisting of four parts: (i) sociodemographic questions, (ii) experiences, (iii) influencing factors, (iv) behavioral intentions. The sample consisted of 309 participants aged between 18 and 70 years.

Results indicate moderately high scores for respondents' opinions regarding perceived severity, performance expectancy, and self-efficacy. When comparing sociodemographic, there was no significant response be concerning age or gender. There are varying degrees of mental health apps used only in terms of country. There is a substantial difference in usage and behavioral intention regarding mental health apps when comparing Germany and Portugal. Additionally, correlation analyses and multiple linear regression demonstrated a meaningful positive relationship between some factors. Finally, the research hypotheses were tested, and it was confirmed that performance expectancy, social influence, hedonic motivation, and action incentives influence the behavioral intention to use a mental health application. This analysis also revealed that participants generally have positive associations with mental health apps and intend to use them in the future.

6.1. THEORETICAL AND MANAGERIAL IMPLICATIONS

Mental health apps are one way to provide smartphone owners with low-cost, hassle-free mental health treatment. Thus, it is an opportunity to counteract the increasing trend of mental illness and close the gap between treatment needs and treatment uptake (MHA, 2021). In this context, they

evaluate the factors that influence the adoption of mental health apps that yields significant results for app providers. Understanding user behavior and the perceptions consumers develop helps brands select the best features and promote their products according to the most effective and influential elements.

The ability to predict intention to use a mental health app helps companies develop marketing strategies because by understanding the psychology of how consumers are influenced to perceive a brand or product; brands can more successfully reach consumers, create positive and influential decision making toward their products, and gain competitive advantage in the marketplace (Lim et al., 2017). This means that suppliers should provide incentives for activities, such as advertising on various platforms. They should choose measures to communicate the effectiveness of these apps and ensure that people are made aware of the issue through campaigns, for example. In addition, they should speak to potential users that using mental health apps is easy and fun to appeal to hedonistic motivation. These factors have the most significant impact on behavioral intention, leading to greater adoption of mental health apps.

6.2. LIMITATIONS AND FURTHER RESEARCH

As with any research, there are several limitations to this study that may lead to recommendations for the future.

First, the sample size is relatively small to represent the population, so the data does not represent the entire target population studied. Therefore the results of this study are not sufficient to generalize the findings. Because data were collected using a random sample and an online questionnaire, most participants were between 20 and 29. Future studies may be of interest to reach a broader age group and further validate whether there are differences in perceptions and influencing factors regarding the adoption of mental health applications. It may also be speculated whether individuals who voluntarily

participate in a mental health survey are generally more open to the topic and are typically part of the population segment more likely to use a mental health app.

In addition, this study did not consider some sociodemographic factors, such as income or education level. Therefore, it would be interesting to investigate possible differences in behavioral intention according to participants' income and education level in future projects. It would also be interesting to analyze the differences in usage between different countries, such as Portugal and Germany, to understand the reasons for this. Moreover, it is essential to note that the data were collected during the COVID 19 pandemic. The respondents in this crisis might have influenced the responses regarding the intention to use a mental health app due to psychological stress. Because the influencing factors and behavioral intention were assessed based on UTAUT2 and HBM factors, some other determinants could not be analyzed. Another limitation is that the study is correlational, so no causation can be concluded.

Finally, examining the long-term use of mental health apps would be essential because of the large gap between initial adoption and long-term service. Therefore, it would be interesting to find out why people stop using these apps and how to counteract this.

7. REFERENCES

- Abbad, M. M. M. (2021). Using the UTAUT model to understand students' usage of e-learning systems in developing countries. *Education and Information Technologies*, *26*(6), 7205–7224. https://doi.org/10.1007/s10639-021-10573-5
- Abotchie, P. N. & Shokar, N. K. (2009). Cervical Cancer Screening Among College Students in Ghana: Knowledge and Health Beliefs. *International Journal of Gynecologic Cancer*, *19*(3), 412–416. https://doi.org/10.1111/igc.0b013e3181a1d6de
- Abraham, C. & Sheeran, P. (2015). The Health Belief Model. *ResearchGate*. https://www.researchgate.net/publication/290193215_The_Health_Belief_Model
- Abrahão, R. D. S., Moriguchi, S. N. & Andrade, D. F. (2016). The intention to adopt mobile payment:
 An analysis in the light of the Unified theory of acceptance and use of technology (UTAUT).
 RAI Revista de Administração e Inovação, *13*(3), 221–230.
 https://doi.org/10.1016/j.rai.2016.06.003
- Adhikari, S. (2019, 22. November). *Health Belief Model- An Interesting Model for Behavior Change!* Public Health Notes. https://www.publichealthnotes.com/health-belief-model-aninteresting-model-for-behavior-change/
- Alraja, M., Hammami, S., Chikhi, B .& Samia, F. (2016). The Influence of Effort and Performance Expectancy on Employees to Adopt E-government: Evidence from Oman. *International Review of Management and Marketing*, 6, 930-934.
- American Psychological Association. (2018). *Stress in America Generation Z.* https://www.apa.org/news/press/releases/stress/2018/stress-gen-z.pdf
- Anastasiadou, D., Folkvord, F., Serrano-Troncoso, E. & Lupiañez-Villanueva, F. (2019). Mobile Health Adoption in Mental Health: User Experience of a Mobile Health App for Patients With an Eating Disorder. *JMIR mHealth and uHealth*, *7*(6), e12920. https://doi.org/10.2196/12920

- Bartosik-Purgat, M., Filimon, N. & Kiygi-Calli, M. (2017). Social Media and Higher Education An International Perspective. *Economics & Sociology*, *10*(1), 181–191. https://doi.org/10.14254/2071-789x.2017/10-1/13
- Bearden, W. O. & Etzel, M. J. (1982). Reference Group Influence on Product and Brand Purchase Decisions. *Journal of Consumer Research*, *9*(2), 183. https://doi.org/10.1086/208911
- Becker, D. (2016). Acceptance of Mobile Mental Health Treatment Applications. *Procedia Computer Science*, *98*, 220–227. https://doi.org/10.1016/j.procs.2016.09.036
- Becker, M. H. (1974). The Health Belief Model and Sick Role Behavior. *Health Education Monographs*, 2(4), 409–419. https://doi.org/10.1177/109019817400200407
- Bell, V. (2007). Online information, extreme communities and internet therapy: Is the internet good for our mental health? *Journal of Mental Health*, *16*(4), 445–457.
 https://doi.org/10.1080/09638230701482378
- Berger, M., Wagner, T. H. & Baker, L. C. (2005). Internet use and stigmatized illness. *Social Science & Medicine*, *61*(8), 1821–1827. https://doi.org/10.1016/j.socscimed.2005.03.025
- Bovin, M. J., Miller, C. J., Koenig, C. J., Lipschitz, J. M., Zamora, K. A., Wright, P. B., Pyne, J. M. &
 Burgess, J. F. (2019). Veterans' experiences initiating VA-based mental health care.
 Psychological Services, *16*(4), 612–620. https://doi.org/10.1037/ser0000233
- Brauser, D. (2020, 25. September). *Suicide, Depression, Anxiety: COVID 19's Heavy Toll on Youth*. Medscape. https://www.medscape.com/viewarticle/938065
- Brown, S. A. & Venkatesh, V. (2005). Model of Adoption of Technology in Households: A Baseline
 Model Test and Extension Incorporating Household Life Cycle. *MIS Quarterly*, *29*(3), 399.
 https://doi.org/10.2307/25148690
- Carmines, E. G. & Zeller, R. A. 1979. Reliability and Validity Assessment, Newbury Park, CA, SAGE.
- Chang, Y. T., Chao, C. M., Yu, C. W. & Lin, F. C. (2021). Extending the Utility of UTAUT2 for Hospital Patients' Adoption of Medical Apps: Moderating Effects of e-Health Literacy. *Mobile Information Systems*, 2021, 1–10. https://doi.org/10.1155/2021/8882317

Cheng, C. & Cheung, M. W. L. (2005). Psychological Responses to Outbreak of Severe Acute
 Respiratory Syndrome: A Prospective, Multiple Time-Point Study. *Journal of Personality*, 73(1), 261–285. https://doi.org/10.1111/j.1467-6494.2004.00310.x

- Chillakuri, B. (2020). Understanding Generation Z expectations for effective onboarding. *Journal of Organizational Change Management*, 33(7), 1277–1296. https://doi.org/10.1108/jocm-02-2020-0058
- Cho, J. (2016). The impact of post-adoption beliefs on the continued use of health apps. *International Journal of Medical Informatics*, *87*, 75–83. https://doi.org/10.1016/j.ijmedinf.2015.12.016
- Clay, R. A. (2021, January). Mental health apps are gaining traction. *Monitor on Psychology*, 52(1). http://www.apa.org/monitor/2021/01/trends-mental-health-apps
- Connolly, S. L., Miller, C. J., Koenig, C. J., Zamora, K. A., Wright, P. B., Stanley, R. L. & Pyne, J. M.
 (2018). Veterans' Attitudes Toward Smartphone App Use for Mental Health Care: Qualitative Study of Rurality and Age Differences. *JMIR mHealth and uHealth*, 6(8), e10748. https://doi.org/10.2196/10748
- Corrigan, P. (2004). How stigma interferes with mental health care. *American Psychologist*, *59*(7), 614–625. https://doi.org/10.1037/0003-066x.59.7.614
- Dabholkar, P. A. (1996). Consumer evaluations of new technology-based self-service options: An investigation of alternative service quality models. *International Journal of Research in Marketing*, *13*(1), 29–51. https://doi.org/10.1016/0167-8116(95)00027-5
- Dodds, W. B., Monroe, K. B. & Grewal, D. (1991). Effects of Price, Brand, and Store Information on Buyers' Product Evaluations. *Journal of Marketing Research*, *28*(3), 307–319. https://doi.org/10.1177/002224379102800305
- Dwivedi, Y. K., Shareef, M. A., Simintiras, A. C., Lal, B. & Weerakkody, V. (2016). A generalized adoption model for services: A cross-country comparison of mobile health (m-health).
 Government Information Quarterly, 33(1), 174–187.
 https://doi.org/10.1016/j.giq.2015.06.003

Education Policy Institute. (2021, 8. April). Social media and children's mental health.

https://epi.org.uk/publications-and-research/social-media-childrens-mental-health-reviewevidence/

El-Terk, N. (2020, 13. März). *Toilet paper, canned food: What explains coronavirus panic buying*. Coronavirus Pandemic News | Al Jazeera.

https://www.aljazeera.com/news/2020/3/13/toilet-paper-canned-food-what-explainscoronavirus-panic-buying

European Commission. (2021). Mental Health.

https://ec.europa.eu/health/non_communicable_diseases/mental_health_en

- Featherman, M. S. & Hajli, N. (2015). Self-Service Technologies and e-Services Risks in Social
 Commerce Era. *Journal of Business Ethics*, *139*(2), 251–269. https://doi.org/10.1007/s10551015-2614-4
- Fricker, R. D., & Schonlau, M. (2002). Advantages and disadvantages of Internet research surveys: Evidence from the literature. *Field methods*, 14(4), 347-367.
- Gao, J., Zheng, P., Jia, Y., Chen, H., Mao, Y., Chen, S., Wang, Y., Fu, H. & Dai, J. (2020). Mental health problems and social media exposure during COVID-19 outbreak. *PLOS ONE*, *15*(4), e0231924. https://doi.org/10.1371/journal.pone.0231924
- Gao, Y., Li, H. & Luo, Y. (2015). An empirical study of wearable technology acceptance in healthcare. Industrial Management & Data Systems, 115(9), 1704–1723. https://doi.org/10.1108/imds-03-2015-0087
- Ghauri, P. N., & Grønhaug, K. (2005). Research Methods in Business Studies: A Practical Guide. London: Pearson Education.
- Glazer, R. (2020, 3. April). *COVID-19 Will Permanently Change The Way Every Generation Lives Here's How*. Forbes. https://www.forbes.com/sites/robertglazer/2020/04/01/covid-19-willpermanently-change-the-way-every-generation-lives-heres-how/?sh=7ab51810493b

- GLSEN. (2019). The 2019 National School Climate Survey. https://www.glsen.org/research/2019national-school-climate-survey
- Goldman, L. (2020, 28. August). *How to recognize the signs of depression and effectively cope with your symptoms*. Insider. https://www.insider.com/what-is-depression
- Grand View Research. (2021). *mHealth Apps Market Size* / *Industry Report, 2021–2028*. https://www.grandviewresearch.com/industry-analysis/mhealth-app-market
- Gupta, B., Dasgupta, S. & Gupta, A. (2008a). Adoption of ICT in a government organization in a
 developing country: An empirical study. *The Journal of Strategic Information Systems*, *17*(2),
 140–154. https://doi.org/10.1016/j.jsis.2007.12.004
- Ha, S. & Stoel, L. (2009). Consumer e-shopping acceptance: Antecedents in a technology acceptance model. *Journal of Business Research*, 62(5), 565–571.
 https://doi.org/10.1016/j.jbusres.2008.06.016
- Hair, J., Black, W., Babin, B., Anderson, R., & Tatham, R. (2018). *Multivariate Data Analysis*. Pearson Prentice Hall.
- Headspace. (2021). About us. https://www.headspace.com/about-us
- Herzog, K. (2020, 24. Mai). *Mental health apps draw wave of new users as experts call for more oversight*. CNBC. https://www.cnbc.com/2020/05/24/mental-health-apps-draw-wave-of-users-as-experts-call-for-oversight.html

Hochbaum, G. M. (1958). Public Participation in Medical Screening Programs. Academic Service.

- Hofstede, G. & Bond, M. H. (1984). Hofstede's Culture Dimensions. *Journal of Cross-Cultural Psychology*, *15*(4), 417–433. https://doi.org/10.1177/0022002184015004003
- Hoonakker, P., & Carayon, P. (2009). Questionnaire survey nonresponse: A comparison of postal mail and internet surveys. *Intl. Journal of Human-Computer Interaction*, 25(5), 348-373.
- Huckvale, K., Torous, J. & Larsen, M. E. (2019). Assessment of the Data Sharing and Privacy Practices of Smartphone Apps for Depression and Smoking Cessation. *JAMA Network Open*, *2*(4), e192542. https://doi.org/10.1001/jamanetworkopen.2019.2542

- Hsu, H. & Chang, Y. (2013). Extended TAM Model: Impacts of Convenience on Acceptance and Use of Moodle. *David Publishing*, 3(4), 211-218.
- Irani, Z., Dwivedi, Y. K. & Williams, M. D. (2009). Understanding consumer adoption of broadband: an extension of the technology acceptance model. *Journal of the Operational Research Society*, 60(10), 1322–1334. https://doi.org/10.1057/jors.2008.100
- Janz, N. K. & Becker, M. H. (1984). The Health Belief Model: A Decade Later. *Health Education Quarterly*, *11*(1), 1–47. https://doi.org/10.1177/109019818401100101
- Jones, C. L., Jensen, J. D., Scherr, C. L., Brown, N. R., Christy, K. & Weaver, J. (2014). The Health Belief Model as an Explanatory Framework in Communication Research: Exploring Parallel, Serial, and Moderated Mediation. *Health Communication*, *30*(6), 566–576. https://doi.org/10.1080/10410236.2013.873363
- Jones, L. V. & Thissen, D. (2006). 1 A History and Overview of Psychometrics. *Handbook of Statistics*, 1–27. https://doi.org/10.1016/s0169-7161(06)26001-2
- Karim, F., Oyewande, A., Abdalla, L. F., Chaudhry Ehsanullah, R. & Khan, S. (2020). Social Media Use and Its Connection to Mental Health: A Systematic Review. *Cureus*. Published. https://doi.org/10.7759/cureus.8627
- Karslidou, N. & Thomas, G. (2021). The effect of body-image, self-esteem, and social media utility on women's sexual satisfaction in Greece..

Kenny, R., Dooley, B. & Fitzgerald, A. (2014). Developing mental health mobile apps: Exploring adolescents' perspectives. *Health Informatics Journal*, 22(2), 265–275. https://doi.org/10.1177/1460458214555041

Kyriazos, T. (2018). Applied psychometrics: sample size and sample power considerations in factor analysis (EFA, CFA) and SEM in general. *Psychology*, 9(08), 2207-2230. https://doi.org/10.4236/psych.2018.98126

- Langley, E. L., Wootton, B. M. & Grieve, R. (2018). The Utility of the Health Belief Model Variables in Predicting Help-Seeking Intention for Anxiety Disorders. *Australian Psychologist*, *53*(4), 291– 301. https://doi.org/10.1111/ap.12334
- Levy, A. (2020, 22. April). *In crisis,, some start-ups are surging and saying "no thanks" to hungry investors*. CNBC. https://www.cnbc.com/2020/04/21/start-ups-surging-in-coronavirus-lockdown-decline-venture-money.html
- Limayem M., Hirt, S. & Cheung, M. K. (2007). How Habit Limits the Predictive Power of Intention: The Case of Information Systems Continuance. *MIS Quarterly*, *31*(4), 705. https://doi.org/10.2307/25148817
- Lipschitz, J., Miller, C. J., Hogan, T. P., Burdick, K. E., Lippin-Foster, R., Simon, S. R. & Burgess, J.
 (2019). Adoption of Mobile Apps for Depression and Anxiety: Cross-Sectional Survey Study on Patient Interest and Barriers to Engagement. *JMIR Mental Health*, 6(1), e11334. https://doi.org/10.2196/11334
- Loiwal, M. (2020, 31. März). 20% increase in patients with mental illness since coronavirus outbreak: Survey. India Today. https://www.indiatoday.in/india/story/20-per-cent-increase-in-patientswith-mental-illness-since-coronavirus-outbreak-survey-1661584-2020-03-31

Ma, Q. & Liu, L. (2005). The Technology Acceptance Model. 10.4018/9781591404743.ch006.ch000.

Malhotra, N. K., Kim, S. S. & Agarwal, J. (2004). Internet Users' Information Privacy Concerns (IUIPC): The Construct, the Scale, and a Causal Model. *Information Systems Research*, *15*(4), 336–355. https://doi.org/10.1287/isre.1040.0032

MarketsandMarkets. (2020). *mHealth Solutions Market by Connected Devices (Glucose & Blood Pressure Monitor, Peak Flow Meter, Pulse Oximeter), Apps (Weight Loss, Women Health, Diabetes Management, Mental Health), Services (Remote Monitoring, Consultation) - Global Forecast to 2025.* https://www.marketsandmarkets.com/Market-Reports/mhealth-apps-andsolutions-market-1232.html

- Marshall, J. M., Dunstan, D. A. & Bartik, W. (2019). Clinical or gimmickal: The use and effectiveness of mobile mental health apps for treating anxiety and depression. *Australian & New Zealand Journal of Psychiatry*, *54*(1), 20–28. https://doi.org/10.1177/0004867419876700
- Marshall, J. M., Dunstan, D. A. & Bartik, W. (2020). Effectiveness of Using Mental Health Mobile Apps as Digital Antidepressants for Reducing Anxiety and Depression: Protocol for a Multiple Baseline Across-Individuals Design. *JMIR Research Protocols*, 9(7), e17159. https://doi.org/10.2196/17159
- Melkert, M., & Vos, K. (2010). A comparison of quantitative and qualitative approaches: complementarities and trade-offs. *Cultural tourism research methods*, 33-40.
- Moser, C. A. & Kalton, G. 1989. Survey methods in social investigation, Aldershot, Gower.
- Michikyan, M. & Suárez-Orozco, C. (2016). Adolescent Media and Social Media Use. *Journal of Adolescent Research*, *31*(4), 411–414. https://doi.org/10.1177/0743558416643801
- Miron, O., Yu, K. H., Wilf-Miron, R. & Kohane, I. S. (2019). Suicide Rates Among Adolescents and Young Adults in the United States, 2000–2017. *JAMA*, *321*(23), 2362.

https://doi.org/10.1001/jama.2019.5054

- Mweshi, G. K. & Sakyi, K. (2020). Application of sampling methods for the research design. *Archives* of Business Research, 8(11), 180–193. https://doi.org/10.14738/abr.811.9042
- National Institude of Mental Health. (2017). *Technology and the Future of Mental Health Treatment*. (NIMH). https://www.nimh.nih.gov/health/topics/technology-and-the-future-of-mentalhealth-treatment
- Newman, M. G., Szkodny, L. E., Llera, S. J. & Przeworski, A. (2011). A review of technology-assisted self-help and minimal contact therapies for anxiety and depression: Is human contact necessary for therapeutic efficacy? *Clinical Psychology Review*, *31*(1), 89–103. https://doi.org/10.1016/j.cpr.2010.09.008

NORC. (2020). COVID Response Tracking Study.

https://www.norc.org/Research/Projects/Pages/covid-response-tracking-study.aspx

OECD (2016). Economiy and social benefits of internet openness. ECONOMIC AND SOCIAL BENEFITS OF INTERNET OPENNESS (oecd.org)

OECD. (2018). 7.B. Workshop: The State of Health in the EU Country Profiles: informing a public health agenda and policy-making. *European Journal of Public Health*, 4(28). https://doi.org/10.1093/eurpub/cky213.591

O'Reilly, M., Dogra, N., Whiteman, N., Hughes, J., Eruyar, S. & Reilly, P. (2018). Is social media bad for mental health and wellbeing? Exploring the perspectives of adolescents. *Clinical Child Psychology and Psychiatry*, 23(4), 601–613. https://doi.org/10.1177/1359104518775154

- Osberg, M. & Mehrotra, D. (2020, 24. Februar). *The Spooky, Loosely Regulated World of Online Therapy*. Jezebel. https://jezebel.com/the-spooky-loosely-regulated-world-of-online-therapy-1841791137
- Park, S. Y. (2009). An Analysis of the Technology Acceptance Model in Understanding University
 Students' Behavioral Intention to Use e-Learning. *Educational Technology & Society*, 12(3), 150-162.
- Parker, L., Halter, V., Karliychuk, T. & Grundy, Q. (2019). How private is your mental health app data?
 An empirical study of mental health app privacy policies and practices. *International Journal of Law and Psychiatry*, 64, 198–204. https://doi.org/10.1016/j.ijlp.2019.04.002
- Phichitchaisopa, N. and Naenna, T. (2013) Factors Affecting the Adoption of Healthcare Information Technology. *EXCLI Journal*, 12, 413-436.

Pfefferbaum, B. & North, C. S. (2020). Mental Health and the Covid-19 Pandemic. *New England Journal of Medicine*, 383(6), 510–512. https://doi.org/10.1056/nejmp2008017

Pikkarainen, T., Pikkarainen, K., Karjaluoto, H. & Pahnila, S. (2004). Consumer acceptance of online banking: an extension of the technology acceptance model. *Internet Research*, 14(3), 224–235. https://doi.org/10.1108/10662240410542652

- Ponchio, M., Barth, N., & Zambaldi, F. (2021). Using the internet for data collection in business research. *RAUSP Management Journal*, 56(2), 254-258. https://doi.org/10.1108/RAUSP-04-2021-251
- Portz, J. D., Bayliss, E. A., Bull, S., Boxer, R. S., Bekelman, D. B., Gleason, K. & Czaja, S. (2019). Using the Technology Acceptance Model to Explore User Experience, Intent to Use, and Use Behavior of a Patient Portal Among Older Adults With Multiple Chronic Conditions:
 Descriptive Qualitative Study. *Journal of Medical Internet Research*, *21*(4), e11604. https://doi.org/10.2196/11604
- Price, M., Yuen, E. K., Goetter, E. M., Herbert, J. D., Forman, E. M., Acierno, R. & Ruggiero, K. J. (2013). mHealth: A Mechanism to Deliver More Accessible, More Effective Mental Health Care. *Clinical Psychology & Psychotherapy*, *21*(5), 427–436. https://doi.org/10.1002/cpp.1855
- Przybylski, A. K., Murayama, K., DeHaan, C. R. & Gladwell, V. (2013). Motivational, emotional, and behavioral correlates of fear of missing out. *Computers in Human Behavior*, 29(4), 1841– 1848. https://doi.org/10.1016/j.chb.2013.02.014
- Ren, S. Y., Gao, R. D. & Chen, Y. L. (2020). Fear can be more harmful than the severe acute respiratory syndrome coronavirus 2 in controlling the corona virus disease 2019 epidemic. *World Journal* of Clinical Cases, 8(4), 652–657. https://doi.org/10.12998/wjcc.v8.i4.652

Research and Markets. (2021, 6. Oktober). Mental Health Apps: Global Market and Competition Outlook 2021–2026. Research and Markets. https://www.prnewswire.com/newsreleases/mental-health-apps-global-market-and-competition-outlook-2021-2026-301394131.html

Rosenstock, I. M. (1974). Historical Origins of the Health Belief Model. *Health Education Monographs*, 2(4), 328–335. https://doi.org/10.1177/109019817400200403

Rössler, W. (2016). The stigma of mental disorders. *EMBO reports*, *17*(9), 1250–1253. https://doi.org/10.15252/embr.201643041

- Scholz, C., Jovanova, M., Baek, E. C. & Falk, E. B. (2020). Media content sharing as a value-based decision. *Current Opinion in Psychology*, *31*, 83–88. https://doi.org/10.1016/j.copsyc.2019.08.004
- Statista. (2021a, August 6). Smartphone users worldwide 2016–2021.

https://www.statista.com/statistics/330695/number-of-smartphone-users-worldwide/ Statista. (2021b, August 31). *Number of active apps from the Apple App Store 2008–2021*.

https://www.statista.com/statistics/268251/number-of-apps-in-the-itunes-app-store-since-2008/

- Statista. (2021b, November 4). Any mental illness in the past year among U.S. adults by age and gender 2020. https://www.statista.com/statistics/252311/mental-illness-in-the-past-year-among-us-adults-by-age-and-gender/
- Stein, D. J., Palk, A. C. & Kendler, K. S. (2021). What is a mental disorder? An exemplar-focused approach. *Psychological Medicine*, *51*(6), 894–901.

https://doi.org/10.1017/s0033291721001185

Stewart, C. (2021, Mai 31). Mental health in Europe - statistics & facts. Statista.

https://www.statista.com/topics/7916/mental-health-in-europe/#topicHeader_wrappe

Taherdoost, H. (2016). Sampling Methods in Research Methodology; How to Choose a Sampling
 Technique for Research. International Journal of Academic Research in Management (IJARM),
 5, 18-27. https://doi.org/10.2139/ssrn.3205035

Taherdoost, H. (2017). A review of technology acceptance and adoption models and theories. *Procedia Manufacturing*, 22, 960–967. https://doi.org/10.1016/j.promfg.2018.03.137

Taherdoost, H. (2020). What Is the Best Response Scale for Survey and Questionnaire Design; Review of Different Lengths of Rating Scale / Attitude, Scale / Likert Scale. *International Journal of Academic Research in Management*, Vol. 8, No. 1, 2020, Page: 1-10.

- Thomas, T., Singh, L. & Gaffer, K. (2013). The utility of the UTAUT model in explaining mobile learning adoption in higher education in Guyana. *International Journal of Education and development*, 9(3), 71-85.
- Thornicroft, G., Brohan, E., Rose, D., Sartorius, N. & Leese, M. (2009). Global pattern of experienced and anticipated discrimination against people with schizophrenia: a cross-sectional survey. *The Lancet*, *373*(9661), 408–415. https://doi.org/10.1016/s0140-6736(08)61817-6
- Venkatesh, V., Morris, M., Davis, G. & Davis, F. (2003). User Acceptance of Information Technology: Toward a Unified View. *MIS Quarterly*, *27*(3), 425. https://doi.org/10.2307/30036540
- Venkatesh, V., Thong, J.& Xu, X. (2012). Consumer Acceptance and Use of Information Technology:
 Extending the Unified Theory of Acceptance and Use of Technology. *MIS Quarterly*, *36*(1), 157. https://doi.org/10.2307/41410412
- Venkatesh, V. & Morris, M. G. (2000). Why Don't Men Ever Stop to Ask for Directions? Gender, Social Influence, and Their Role in Technology Acceptance and Usage Behavior. *MIS Quarterly*, 24(1), 115. https://doi.org/10.2307/3250981
- Verma, J., & Verma, P. (2020). Use of G* Power Software. In Determining Sample Size and Power in Research Studies (pp. 55-60). Springer.
- Wang, C., Pan, R., Wan, X., Tan, Y., Xu, L., Ho, C. S. & Ho, R. C. (2020). Immediate Psychological Responses and Associated Factors during the Initial Stage of the 2019 Coronavirus Disease (COVID-19) Epidemic among the General Population in China. *International Journal of Environmental Research and Public Health*, *17*(5), 1729. https://doi.org/10.3390/ijerph17051729

Watson, A. C., Corrigan, P., Larson, J. E. & Sells, M. (2006). Self-Stigma in People With Mental Illness. Schizophrenia Bulletin, 33(6), 1312–1318. https://doi.org/10.1093/schbul/sbl076

 Weiner, B., Perry, R. P. & Magnusson, J. (1988). An attributional analysis of reactions to stigmas.
 Journal of Personality and Social Psychology, 55(5), 738–748. https://doi.org/10.1037/0022-3514.55.5.738 World Health Organization. (2012). Risks to mental health: an overview of vulnerabilities and risk

factors.

https://www.who.int/mental_health/mhgap/risks_to_mental_health_EN_27_08_12.pdf#:~:t ext=At%20its%20Sixty-

fifth%20session%2C%20held%20in%20May%202012%2C,a%20basis%20for%20developing% 20the%20mental%20health%20plan.

World Health Organization. (2019). Mental disorders. https://www.who.int/news-

room/factsheets/detail/mental-disorders

World Health Organization. (2020). Mental health and COVID-19.

https://www.euro.who.int/en/health-topics/health-emergencies/coronavirus-covid-19/publications-and-technical-guidance/noncommunicable-diseases/mental-health-andcovid-19

- World Health Organization, Victorian Health Promotion Foundation. Department of Mental Health and Substance Abuse & University of Melbourne. (2004). *Promoting Mental Health*. https://www.who.int/publications-detail-redirect/9241562943
- Yuan, S., Ma, W., Kanthawala, S. & Peng, W. (2015). Keep Using My Health Apps: Discover Users'
 Perception of Health and Fitness Apps with the UTAUT2 Model. *Telemedicine and e-Health*, 21(9), 735–741. https://doi.org/10.1089/tmj.2014.0148
- Yueh, H. P., Huang, J. Y. & Chang, C. (2015). Exploring factors affecting students' continued Wiki use for individual and collaborative learning: An extended UTAUT perspective. *Australasian Journal of Educational Technology*, 31(1). https://doi.org/10.14742/ajet.170

8. APPENDIX

Published survey

By participating in this survey you will help me to finish my master's thesis. Thank you very much! :)

Informed consent By clicking "I agree", I declare that I am at least 18 years old and that I agree to participate in this research. I declare that I have been informed that my participation in this study is voluntary, that I can leave the survey at any time without penalty, and that all data will be kept confidential. I understand that there are no serious risks associated with this study.

 \bigcirc I do not agree to participate in this survey (1)

I agree to participate in this survey (2)

Skip To: End of Survey If By clicking "I agree", I declare that I am at least 18 years old and that I agree to participate... = I do not agree to participate in this survey

End of Block: Informed consent

Start of Block: Introduction

Introduction Mental health apps like Headspace, Calm, or Moodfit focus on improving various aspects of mental health and well-being. They meet a range of needs and offer different functions. Some focus on meditation, mood tracking, anti-stress exercises and sleep aids, while others serve more as a diary or online therapy where you can speak with a licensed mental health professional. Apart from a few paid apps, most of them can be downloaded and used for free by any smartphone user.

End of Block: Introduction

Start of Block: Demographics



Age What's your age? (full numbers)

Gender What's your gender?	
O Female (1)	
\bigcirc Male (2)	
O Non-binary / third gender (3)	
\bigcirc Prefer not to say (4)	
$\chi \rightarrow$	

Country Which country are you from?

▼ Afghanistan (1) ... Zimbabwe (1357)

End of Block: Demographics

Start of Block: Experience

Usage past Have you ever used a mental health app?

O No (1)

O Yes (2)

91

Usage present Are you currently using a mental health app?

No (27)
 Yes (28)
 Page Break

Display This Question:

If Are you currently using a mental health app? = Yes

Name app Please name the mental health app/ apps you are currently using.

End of Block: Experience

Start of Block: Performacy expectancy

PE Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
Using a mental health app would improve my mental wellbeing. (1)	0	0	0	0	0
Using a mental health app can satisfy my mental health care needs. (2)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using a mental health app can improve my efficacy in monitoring my mental health conditions. (3)	\bigcirc	0	\bigcirc	0	\bigcirc
Using a mental health app will improve the quality of my life. (4)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

End of Block: Performacy expectancy

Start of Block: Effort expectancy

EE Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
Choosing a mental health app would require a lot of effort. (1)	0	0	0	0	0
Using a mental health app would require a lot of effort. (2)	0	0	\bigcirc	\bigcirc	\bigcirc
Using a mental health app would require a lot of time. (3)	\bigcirc	0	\bigcirc	0	0
Learning how to use a mental health app would be difficult. (4)	0	0	0	\bigcirc	0

End of Block: Effort expectancy

Start of Block: Social influence

SI Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
People have already asked me about my mental health. (1)	0	0	0	0	0
People have already advised me to seek mental health help. (2)	\bigcirc	\bigcirc	0	\bigcirc	0
People have already advised me to use a mental health app. (3)	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc
I					

End of Block: Social influence

Start of Block: Hedonic motivation

HM Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
I think using a mental health app is fun. (1)	0	0	0	0	0
I think using a mental health app is entertaining. (2)	0	0	0	\bigcirc	0
l think using a mental health app is enjoyable. (3)	0	\bigcirc	0	\bigcirc	0

End of Block: Hedonic motivation

Start of Block: Perceived susceptibility

PSU Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
Everybody can get a mental health disease. (1)	0	0	0	0	0
l am not at risk of mental health disease. (2)	0	0	0	\bigcirc	\bigcirc
I can have a mental health disease even without feeling its signs and symptoms (3)	0	0	\bigcirc	\bigcirc	\bigcirc
I am afraid of getting a mental health disease. (4)	0	0	0	\bigcirc	0

End of Block: Perceived susceptibility

Start of Block: Perceived severity

PSE Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
Mental health diseases can lead to death. (1)	0	0	0	0	0
Mental health diseases can change the whole life. (2)	0	0	0	0	0
Mental health diseases can disrupt the harmony in families. (3)	0	\bigcirc	0	\bigcirc	0
Mental health diseases are long lasting. (4)	\bigcirc	0	\bigcirc	\bigcirc	0

End of Block: Perceived severity

Start of Block: Self-efficacy

SE Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
I can always manage to solve difficult problems if I try hard enough. (1)	0	0	0	0	0
It is easy for me to accomplish my goals. (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I am confident that I could deal efficiently with unexpected events (3)	0	0	0	\bigcirc	0

End of Block: Self-efficacy

Start of Block: Cues to action

CA Please rate the following statements.

	No (32)	Maybe (33)	Yes (34)
I heard about mental health apps before. (1)	0	0	0
I saw an advertisement about a mental health app. (3)	0	\bigcirc	\bigcirc
I know at least one person who is using a mental health app. (5)	0	\bigcirc	0
Someone has already recommended a mental health app to me. (6)	0	\bigcirc	\bigcirc

End of Block: Cues to action

Start of Block: Behavioral intention

BI Please rate the following statements.

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
l intend to use a mental health app. (1)	0	0	0	\bigcirc	0
l intend to check the availability of a suited mental health app. (2)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
l plan to use a mental health app. (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
It is worth using a mental health app. (4)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

End of Block: Behavioral intention